SCHEDULE 4 AMENDED AND RESTATED OPERATION AGREEMENT

AMENDED AND RESTATED OPERATION AGREEMENT

by and among

UNION ELECTRIC COMPANY d/b/a AmerenUE

CENTRAL ILLINOIS PUBLIC SERVICE COMPANY d/b/a AmerenCIPS

AMERICAN TRANSMISSION SYSTEMS, INCORPORATED

NORTHERN INDIANA PUBLIC SERVICE COMPANY

and

GRIDAMERICA LLC

FEBRUARY 14, 2003

TABLE OF CONTENTS

ARTICLE I	DEFINITIONS	2
1.1. 1.2.	DefinitionsRules of Construction	
ARTICLE II	AUTHORIZATIONS BY THE TRANSMISSION OWNERS	12
2.1.	Functional Control Authorization.	12
2.2.	Non-transferred Facilities	13
2.3.	Interconnection Agreements	
2.4.	Network Upgrades	
2.5.	Revenue Collection and Distribution.	
2.6.	The Company as Owner of Transmission Facilities; Non-Discrimination	
ARTICLE III	RIGHTS, POWERS, AND OBLIGATIONS OF THE GridAmerica	. 16
2.4		
3.1.	Operation And Planning	
3.2	Other Matters	
3.3.	Responsibilities To The Transmission Owner.	
3.4.	Additional Obligations	
3.5.	Information	24
ARTICLE IV	RIGHTS, POWERS, AND OBLIGATIONS OF THE	
	TRANSMISSIONS OWNERS	24
4.1.	Operation and Planning	24
4.2.	Additional Obligations	24
4.3.	Payments to the Company	24
4.4.	Transmission Owners' Remedies	35
ARTICLE V	TERM AND TERMINATION	36
5.1.	Term	36
5.2.	Termination	38
5.3.	Effect of Termination Pursuant to Sections 5.1 or 5.2.3	39
5.4.	Effect of Termination Pursuant to Section 5.2.2	40
5.5.	Regulatory And Other Approvals Or Procedures	41
ARTICLE VI	DISPUTE RESOLUTION	41
6.1.	Negotiations	41
6.2.	Arbitration	41
6.3.	Arbitration of Certain Claims Regarding Removal of Managing Member	46

ARTICLE VII	TAX MATTERS	48
7.1.	Responsibility for Transmission Owner Taxes	48
7.2.	Responsibility for the Company Taxes	
ARTICLE VIII	FORCE MAJEURE	49
ARTICLE IX	BREACH, CURE AND DEFAULT	49
9.1.	Continued Operation	49
9.2.	Remedies	49
9.3.	No Joint Liability	50
ARTICLE X	MISCELLANEOUS PROVISIONS	50
10.1.	Not for Benefit of Third Parties	50
10.2.	Governing Law	50
10.3	Successors And Assigns	
10.4	Effect of Waiver	51
10.5	Severability	51
10.6	Renegotiation	
10.7	Representation And Warranties	51
10.8	Further Assistance	
10.9	Notices	53
10.10	Regulatory Proceedings	53
10.11	Entire Agreement; Amendments	
10.12	Efforts of the Parties	53
10.13	Third-Party Joint Agreements	54
10.14	Confidentiality	54
10.15	No Partnership	57
10.16	Current Documents	57
10.17	Late Payments	57
10.18	Counterparts	57
10.19	Attorneys' Fees	57
10.20	Time is of the Essence	
10.21	Representatives	57

APPENDICES

APPENDIX A DESCRIPTION OF TRANSFERRED FACILITIES

APPENDIX B THIRD-PARTY JOINT AGREEMENTS

APPENDIX C NOTICES

APPENDIX D - SCHEDULE 5A - DELINEATION OF FUNCTIONS BETWEEN MIDWEST ISO, GRIDAMERICA AND GRIDAMERICA THREE

AMENDED AND RESTATED OPERATION AGREEMENT

THIS AMENDED AND RESTATED OPERATION AGREEMENT is made and entered into as of the 14th day of February, 2003 by and among (i) UNION ELECTRIC COMPANY d/b/a AmerenUE and CENTRAL ILLINOIS PUBLIC SERVICE COMPANY d/b/a AmerenCIPS (collectively, "Ameren," and for all purposes of this Agreement, a single Transmission Owner), AMERICAN TRANSMISSION SYSTEMS, INCORPORATED ("ATSI"), and NORTHERN INDIANA PUBLIC SERVICE COMPANY ("NIPSCO"), (ii) GRIDAMERICA LLC, a Delaware limited liability company (the "Company"), each of which may be referred to as a "Party", or collectively as the "Parties."

RECITALS

WHEREAS, the United States Federal Energy Regulatory Commission (together with any successor agency, the "Commission") in Order No. 2000 called for the formation of regional transmission organizations to promote the creation of large electricity markets and to provide reliable, cost-efficient services to customers;

WHEREAS, the Midwest Transmission System Operator, Inc. ("<u>Midwest ISO</u>") is a Commission-approved regional transmission organization.

WHEREAS, on April 25, 2002, the Commission issued an order in Docket No. EL02-65 (99 FERC ¶ 61,105 (2002)) encouraging the formation of an Independent Transmission Company within the Midwest ISO.

WHEREAS, Ameren, ATSI and NIPSCO wish to comply with Order No. 2000 through the formation of an Independent Transmission Company within the Midwest ISO.

WHEREAS, on October 31, 2002, (i) Ameren, ATSI and NIPSCO (or their applicable affiliates), National Grid USA ("NGUSA"), the Initial Member and the Company entered into a Master Agreement dated as of October 31, 2002 (the "Original Master Agreement"), (ii) the Initial Member entered into the Limited Liability Company Agreement of the Company dated as of October 31, 2002 (the "Original LLC Agreement"), (iii) the Company and the Original GridAmerica Companies, or their applicable affiliates, entered into the Operation Agreement dated as of October 31, 2002 (the "Original Operation Agreement") and (iv) the Company and the Midwest ISO entered into the Appendix I ITC Agreement dated as of October 31, 2002 (the "Original MISO ITC Agreement");

WHEREAS, on December 19, 2002, the Commission issued an order in Docket Nos. ER02-2233-001 and EC03-14-000 (101 FERC ¶ 61,320 (2002)) (the "FERC Approving Order") conditionally accepting for filing, suspending and making effective subject to future refund, future filings and further orders the Original Master Agreement, the Original LLC Agreement, the Original Operation Agreement and the Original MISO ITC Agreement;

WHEREAS, each Transmission Owner agrees to transfer Functional Control over its Transmission Facilities to the Company and desires the Company to exercise Functional Control

over its Transferred Facilities on the terms and conditions set forth herein and in the MISO ITC Agreement;

WHEREAS, the Company agrees to accept Functional Control over the Transferred Facilities of the Transmission Owners on the terms and conditions set forth herein and in the MISO ITC Agreement, in each case as modified in compliance with the FERC Approving Order.

NOW THEREFORE, in consideration of the premises, and the mutual representations, warranties, covenants, and agreements hereinafter set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound, each Transmission Owner, each acting in its individual capacity, and the Company agree to amend and restate the Original Operation Agreement in its entirety as follows:

ARTICLE I DEFINITIONS

- **1.1 Definitions.** Terms with initial capitalization used in this Agreement without other definition shall have the meanings specified in this <u>Article I</u>.
 - "AAA" shall have the meaning given in Section 6.2(a).
- "Acceptable Credit Bank" shall mean a bank that (i) is subject to review and examination by a federal Governmental Authority, (ii) is in good standing with such authority, (iii) has combined capital, surplus and undivided profits aggregating not less than \$500 million and (iv) has unsecured long-term debt rated at least "A-" by Standard and Poor's Ratings Group and "A3" by Moody's Investors Service.
- "<u>Accounting Failure</u>" shall mean a material deficiency in the Company's accounting practices or systems as identified in an audit conducted pursuant to the terms of this Agreement.
 - "Additional Arbitration Request" shall have the meaning given in Section 6.2(i).
 - "Additional Claim" shall have the meaning given in Section 6.2(i).
 - "Additional Term" shall have the meaning given in Section 5.1.
- "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly Controlling, Controlled by or under common Control with such Person. As used in this definition, "Control" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise); provided, however, that, in any event, any Person that owns directly or indirectly securities having at least a majority of the voting power for the election of directors or other members of the governing body of a corporation or at least a majority of the partnership or other ownership interests (that carry voting power) of any other Person will be deemed to Control such corporation or other Person.

- "<u>Agreement</u>" shall mean this Amended and Restated Operation Agreement dated as of February 14, 2003, as it may be amended, modified or otherwise supplemented and in effect from time to time.
 - "Ameren" shall have the meaning given in the preamble.
 - "Ancillary Services" shall have the meaning given in the MISO OATT.
- "Applicable Laws and Regulations" shall mean all applicable federal, state and local Laws, ordinances, rules and regulations, and all duly promulgated orders and other duly authorized actions of any Governmental Authority having jurisdiction over a Party, its facilities and/or the services it provides.
 - "Approved Uses" shall have the meaning given in Section 10.14(a)(ii).
 - "Arbitration" shall have the meaning given in Section 6.2.
 - "Arbitration Notice" shall have the meaning given in Section 6.2(b).
 - "Arbitration Rules" shall have the meaning given in Section 6.2(a).
 - "ATSI" shall have the meaning given in the preamble.
- "Business Day" shall mean any day other than Saturday, Sunday, or other day on which banks are authorized or required to be closed in New York, New York.
- "Cause" shall mean (i) Gross Negligence that causes, or is reasonably likely in the future to cause, a Material Adverse Effect, (ii) Willful Misconduct that causes, or is reasonably likely in the future to cause, a Material Adverse Effect or (iii) in the case of the Initial Member as Managing Member (A) the occurrence of any two Counted Years during any five calendar year period, or (B) the failure of NGUSA to comply in any material respect with any of its obligations set forth in Article III or Section 10.1 of the Master Agreement.
 - "Claimant Party" shall have the meaning given in Section 6.2(b).
 - "Claims" shall have the meaning given in Section 6.2(a).
 - "Collection Account" shall have the meaning given in Section 3.3.4(a).
- "Commission" or "FERC" shall mean the Federal Energy Regulatory Commission or any successor agency.
 - "Company Payments" shall have the meaning given in Section 3.4.1(b).
- "Company" shall mean GridAmerica LLC and any wholly-owned subsidiaries created for the purpose of satisfying state domestication requirements.
- "<u>Confidential Information</u>" shall mean all confidential or trade secret information of a Disclosing Party provided to a Recipient pursuant to or in connection with any Transaction

Agreement, including business information; strategies; methods; technical information; pricing techniques and strategies; customer information; investor information; price curves; positions, plans and strategies for expansion or acquisitions, budgets, customer lists, studies of information and data, electronic databases, computer programs, bids or proposals, organizational structure, compensation of personnel, and new product information; provided, however, "Confidential Information" shall not include information that (i) was already known by (as established by dated documentation) a Recipient at the time of the receipt of such information by such Recipient from the Disclosing Party, (ii) is in, or subsequently enters, the public domain other than as a result of a disclosure by the Recipient in breach of an obligation of confidence, (iii) is received by the Recipient from a third party if such third party was not known to be subject to any confidentiality obligation, (iv) is independently developed by a Person without access to the Confidential Information provided by the Disclosing Party, (v) was or is furnished by a Disclosing Party to another Person without written confidentiality restrictions or (vi) is approved for release by written authorization of the Disclosing Party.

"Consent" shall mean any authorization, consent, opinion, order, approval, license, franchise, ruling, permit, tariff, rate, certification, exemption, filing or registration from, by, or with any Governmental Authority, any Person or any governing body of any Person.

"Counted Year" shall mean (i) any calendar year in which the Managing Member would have had liability under Section 11.8(e)(i) of the LLC Agreement but for the application of the limitation contained in clause (ii) of the proviso to such Section 11.8(e)(i) or (ii) any calendar year that is an Operation Agreement Counted Year; provided, however, that there may be only one Counted Year in any calendar year.

"Disclosing Party" shall have the meaning given in Section 10.14.

"Dispute Parties" shall have the meaning given in Section 6.2(b).

"Early Termination Event" shall have the meaning given in the LLC Agreement.

"Effective Date" shall mean October 31, 2002.

"Emergency" shall mean an event or situation which poses an imminent threat of material damage to property or injury (including death) to persons, or is imminently likely to cause a material adverse effect on the security of the Transmission System or the electrical or transmission systems of any other Person interconnected to the Transmission System.

"Entity" shall mean a corporation, limited liability company, partnership, limited partnership, trust, firm, association, or other organization which has a legal existence under the Laws of its jurisdiction of formation which is separate and apart from its owner or owners and any Governmental Authority.

"Fair Market Value" shall have the meaning given in the Master Agreement.

"FERC Approving Order" shall have the meaning given in the recitals hereof.

"Force Majeure" shall have the meaning given in the MISO OATT.

"<u>Functional Control</u>" shall mean the exercise by the Company of control over the operation of the Transmission System and performance of all of the activities contemplated by Order 2000. Notwithstanding anything to the contrary in this Agreement, Functional Control shall include all activities now or hereafter required under Applicable Laws and Regulations to be performed by RTOs.

"Generator" shall mean an Entity that owns or controls and operates an electric power generation facility which produces electrical energy.

"Good Business Practice" shall mean (i) the Company's exercise of Functional Control over the Transferred Facilities, except as ceded to the Midwest ISO under the MISO ITC Agreement, in accordance with Good Utility Practice and (ii) the Company's fulfillment in a commercially reasonable manner and, where applicable, in accordance with Good Utility Practice, of its obligations hereunder and under the other Transaction Agreements, all Applicable Laws and Regulations and all other agreements to which the Company is a party.

"Good Business Practice Breach" shall have the meaning given in Section 3.4.1(a).

"Good Utility Practice" shall have the meaning given in the MISO OATT.

"Governmental Authority" shall mean a federal, state, local or foreign governmental authority; a state, province, commonwealth, territory or district thereof; a county or parish; a city, town, township, village or other municipality; a district, ward or other subdivision of any of the foregoing; any executive, legislative or other governing body of any of the foregoing; any agency, authority, board, department, system, service, office, commission, committee, council or other administrative body of any of the foregoing; any court or other judicial body and any officer, official or other representative of any of the foregoing.

"GridAmerica HoldCo" shall have the meaning given in the Master Agreement.

"GridAmerica ITC" shall mean the ITC created by the Transmission Owners, the Company and NGUSA pursuant to the Master Agreement, the LLC Agreement and this Agreement.

"Gross Negligence" shall mean the gross negligence of (i) the Company in the performance of its duties or obligations under this Agreement other than in its capacity as an owner of facilities that comprise part of the Transmission System or (ii) any Affiliate of the Company that provides services to or for the benefit of the Company in the performance of those services, other than any Affiliate which owns facilities that are part of the Transmission System in its capacity as such.

"<u>Indemnified Owners</u>" shall have the meaning given in <u>Section 4.2.4(a)</u>.

"Indemnified Party" shall have the meaning given in Section 4.2.5(a).

"Indemnifying Owner" shall have the meaning given in Section 4.2.4(a).

"Indemnity Cap" shall have the meaning given in Section 4.2.4(c).

"Indemnifying Party" shall have the meaning given in Section 4.2.5(a).

"Independent Transmission Company" or "ITC" shall mean an independent transmission company approved pursuant to Commission order or regulation.

"<u>Initial Management Fee</u>" shall have the meaning given on the LLC Agreement.

"<u>Initial Member</u>" shall have the meaning given in the LLC Agreement.

"<u>Initial Public Offering</u>" shall have the meaning given in the LLC Agreement.

"<u>Initial Term</u>" shall mean the period commencing on the Effective Date and ending on the fifth anniversary of the Transmission Service Date.

"Interconnection Agreement" shall mean an agreement between the Company and either (i) a Generator governing the terms and conditions of the interconnection of a generation facility to the Transmission System or (ii) a local distribution entity governing the terms and conditions of the interconnection of distribution facilities to the Transmission System.

"<u>Interconnection Customer</u>" shall mean a Generator or local distribution entity which has entered into an Interconnection Agreement.

"Interconnection Procedures" shall mean those procedures and form of agreement governing the interconnection of the facilities of a Generator or a local distribution entity with the Transferred Facilities which are established by the Company and effective pursuant to Applicable Laws and Regulations or, prior to such effectiveness, the MISO OATT and the interconnection protocols of the Midwest ISO, as each may be amended, modified or otherwise supplemented from time to time.

"Interconnection Service" shall, with respect to a Generator, have the meaning given in the MISO OATT, and, with respect to a local distribution entity, have the meaning given in the Interconnection Agreement between the Company and such local distribution entity.

"ITC Order" shall mean the order issued by the Commission authorizing the Company to operate as an ITC within the Midwest ISO pursuant to the terms of the MISO ITC Agreement and any other order of the Commission pertaining to the Company's rights or responsibilities with respect to the Transferred Facilities.

"<u>Law</u>" shall mean any applicable constitutional provision, statute, act, code (including the United States Internal Revenue Code of 1986, as amended from time to time), law, regulation, rule, ordinance, order, decree, ruling, proclamation, resolution, judgment, decision, declaration, or interpretive or advisory opinion of a Governmental Authority.

"<u>Liability Cap Amount</u>" shall mean, in any calendar year, an amount equal to the Initial Management Fee for such calendar year.

"<u>LLC Agreement</u>" shall mean the Amended and Restated Limited Liability Company Agreement of the Company dated as of February 14, 2003, as the same may be amended, modified, or otherwise supplemented and in effect from time to time.

"Lockbox Account" shall have the meaning given in Section 3.3.4(a).

"Lockbox Bank" shall have the meaning given in Section 3.3.4(a).

"Lockbox Subaccount" shall have the meaning given in Section 3.3.4(a)(ii).

"<u>Losses</u>" shall mean any and all damages, losses, claims, demands, suits, recoveries, costs, expenses, liabilities to third parties, reasonable attorneys' fees, and penalties or other sanctions imposed by Governmental Authorities.

"<u>Majority of Indemnifying Transmission Owners</u>" shall have the meaning given in Section 4.2.5(b).

"Management Fee" shall have the meaning given in the LLC Agreement.

"Managing Member" shall mean the managing member of the Company as designated in accordance with Section 6.1 of the LLC Agreement.

"Market Participant" shall mean a Person that is a "Market Participant" within the meaning of Order 2000, or any subsequent rule, regulation or order of the Commission establishing the requirements of independence for a Person managing an ITC exercising the functions and responsibilities that the Company will exercise under the MISO ITC Agreement.

"Master Agreement" shall mean the Amended and Restated Master Agreement dated as of February 14, 2003 among the Company, NGUSA and each Transmission Owner or its applicable Affiliate as the same may be amended, modified or otherwise supplemented and in effect from time to time.

"<u>Material Adverse Effect</u>" means an effect that is, or is reasonably likely to be, materially adverse to the business, assets, condition (financial or otherwise) or operations of the Transmission System taken as a whole.

"Member" shall mean any Person who is a member of the Company, including the Managing Member.

"Midwest ISO" shall mean the Midwest Independent Transmission System Operator, Inc.

"MISO ITC Agreement" shall mean the Amended and Restated Appendix I ITC Agreement by and between the Midwest ISO and the Company dated as of February 14, 2003, as the same may be amended, modified or otherwise supplemented and in effect from time to time.

"<u>MISO OATT</u>" shall mean the Open Access Transmission Tariff of the Midwest Independent Transmission System Operator, Inc. on file with the Commission, as it may be amended, modified or otherwise supplemented and in effect from time to time.

"NERC" shall mean the North American Reliability Council, applicable regional electric reliability councils, or successor organizations.

"Net Plant" or "net plant" shall mean, as of the date of determination thereof and with respect to any Transmission Facilities, the net book value of such Transmission Facilities as computed using the information shown in the then most recent FERC Form 1 filed with the Commission with respect to such Transmission Facilities. For the avoidance of doubt, for any and all purposes of this Agreement and the other Transaction Agreements, (i) "Net Plant" shall be calculated, and if required adjusted, annually on each anniversary of the Effective Date and (ii) the calculation made and FERC Form 1 information used shall be the difference between (A) the information on page 207, Electric Plant in Service (Account 101, 102, 103 and 106), line 53, Total Transmission Plant, Column G, less (B) the information on page 219, Accumulated Provision for Depreciation of Electric Utility Plant (Account 108), Section B. Balances at End of Year According to Functional Classification, line 23, Transmission, Column C; provided, however, that if FERC Form 1 is modified or changed such that the foregoing designations no longer apply, the information used shall be that information in the modified or changed form that provides, as nearly as practicable, the same substantive result as the foregoing.

"Network Upgrades" shall have the meaning given in the MISO OATT.

"NGUSA" shall mean National Grid USA.

"NIPSCO" shall have the meaning given in the preamble.

"Non-Market Participant" shall mean a Person that is not a Market Participant.

"Non-transferred Facilities" shall mean, with respect to any Transmission Owner, such Transmission Owner's transmission facilities and distribution facilities (i) which are not Transferred Facilities, but which are necessary for the provision of Transmission Service or Wholesale Distribution Service to Eligible Customers pursuant to the MISO OATT and which may be subject to an agency agreement; and that are disclosed to the Company prior to the Transmission Service Date or (ii) as a Transmission Owner may subsequently include as "Non-transferred Facilities" for purposes of this Agreement.

"Notice of Removal Dispute" shall have the meaning given in Section 6.3.2.

"Operation Agreement Counted Year" shall mean a calendar year with respect to which either (i) any Transmission Owner shall have any indemnity obligation to the Company pursuant to Section 4.2.4(c) with respect to any Good Business Practice Breach or (ii) any Transmission Owner would have had a claim for indemnification against the Company pursuant to Section 3.4.1(a) with respect to any claim occurring in such year and involving any Good Business Practice Breach but for the fact that the amount of Losses suffered or incurred by such Transmission Owner with respect to such claim plus the Company Payments in respect of Losses occurring in such year exceeded the Liability Cap Amount for such year.

"Operational Failure" shall mean in any calendar year (a) the failure of the Company to perform the functions it is permitted to perform under the MISO ITC Agreement with respect to the delivery of more than (i) 0.01% of the total megawatt hours of Transmission Transactions

within the Transmission System or (ii) 0.4% of the total megawatt hours of approved Transmission Transactions for a particular contract path within the Transmission System or (b) any damage to facilities that comprise part of any Transmission Owner's Transferred Facilities caused solely by the approval by the Company to the extent permitted under the MISO ITC Agreement of a volume of Transmission Transactions that exceed the capacity of the affected facilities; provided, however, that failures caused by (i) Force Majeure occurrences, (ii) the failure of a Transmission Owner to follow Good Utility Practice or (iii) circumstances arising on an interconnected transmission system shall not be considered in determining whether an Operational Failure has occurred.

"Order 2000" shall mean the Commission's order identified as Regional Transmission Organizations, Docket No. RM99-2-000, 89 FERC \P 61,285 (1999), all subsequent orders of the Commission in such docket, and all other orders of the Commission pertaining to the rights and obligations of an RTO.

"Overpaid Party" shall have the meaning given in Section 3.3.4(c).

"Overpayment" shall have the meaning given in Section 3.3.4(c).

"Panel" shall have the meaning given in Section 6.2(d).

"Party" shall have the meaning given in the preamble.

"Pass-Through Basis" shall mean that the obligation of a Party making a payment to another Party under this Agreement shall be only to pay that amount which such Party receives from a third party in respect of such payment obligation to such Person.

"Payment Event" shall have the meaning given in Section 4.2.4(c).

"Percentage Interests" shall have the meaning given in the LLC Agreement.

"Person" shall mean any natural person or Entity.

"Recipient" shall have the meaning given in Section 10.14.

"Regional Transmission Organization" or "RTO" shall have the meaning given in 18 C.F.R § 35.34(b)(1) of the Commission's regulations or such successor definition approved by the Commission.

"Related Proceeding" shall have the meaning given in Section 6.2(c).

"Removal Arbitration" shall have the meaning given in Section 6.3.

"Removal Claim" shall have the meaning given in Section 6.3.

"Removal Claimant" shall have the meaning given in Section 6.3.2.

"Removal Dispute Parties" shall have the meaning given in Section 6.3.2.

"Removal Notice" shall have the meaning given in Section 6.3.2.

"Removal Respondent Party" shall have the meaning given in Section 6.3.2.

"Representatives" shall have the meaning given in Section 10.14(a).

"Respondent Party" shall have the meaning given in Section 6.2(b).

"SEOs" shall have the meaning given in Section 6.1.

"Super Majority of Transmission Owners" shall mean (i) prior to the date on which the Company first issues Units in exchange for Transmission Facilities, two-thirds or more of the Transmission Owners and (ii) thereafter, one or more owners of transmission facilities who, among them, own (through actual or deemed ownership as provided below) Transmission Facilities that are subject to the Functional Control of the Company pursuant to this Agreement or are owned by the Company with a Net Plant greater than 66.67% of the aggregate Net Plant of all Transmission Facilities subject to such Functional Control of the Company pursuant to this Agreement or are owned by the Company. For purposes of the above vote, the "owner of transmission facilities" means (i) in the case of Transmission Facilities subject to the Company's Functional Control pursuant to this Agreement, the Person that actually owns such Transmission Facilities and (ii) in the case of Transmission Facilities actually owned by the Company, the Members in accordance with their respective Percentage Interests (as defined in the LLC Agreement). In the event that an Initial Public Offering shall have occurred, the independent board members of GridAmerica HoldCo shall vote the deemed ownership interest of GridAmerica HoldCo.

"Taxes" shall mean all taxes, charges, fees, levies, penalties, and all other assessments imposed by any Governmental Authority, including, but not limited to, income, excise, property sales, transfer, franchise, payroll, withholding, social security, or other taxes, including any interest, penalties, or additional charges attributable thereto.

"<u>Tax Return</u>" shall mean any return, report, information return or other document (including any related or supporting information) required to be filed with or supplied to any Governmental Authority with respect to Taxes.

"Third Party Claims" shall have the meaning given in Section 4.2.4(c).

"Third Party Recipient" shall have the meaning given in Section 10.14(c).

"Transaction Agreements" shall have the meaning given in the LLC Agreement.

"Transferred Facilities" shall mean, with respect to any Transmission Owner, the Transmission Facilities owned by such Transmission Owner, for which Functional Control has been transferred to the Company pursuant to this Agreement and in compliance with Applicable Laws and Regulations, and which are described on such Transmission Owner's subappendix to Appendix A, as Appendix A may be amended, modified or otherwise supplemented from time to time in compliance with Applicable Laws and Regulations.

"<u>Transmission Business</u>" shall mean, with respect to a Party, its business, assets, and activities relating to the transmission of electricity through Transferred Facilities.

"Transmission Customer" shall have the meaning given in the MISO OATT.

"<u>Transmission Facilities</u>" shall mean facilities used for the transmission of electric power and energy of the kind subject to the jurisdiction of the Commission.

"<u>Transmission Owner</u>" shall mean a Person who is a Party to this Agreement and transfers Functional Control of Transmission Facilities to the Company. The initial Transmission Owners are Ameren, ATSI, NIPSCO.

"<u>Transmission Service</u>" shall mean "Transmission Service" as defined in the MISO OATT and "Network Integration Transmission Service" as defined in the MISO OATT but shall not include Interconnection Service.

"Transmission Service Date" shall have the meaning given in the Master Agreement.

"<u>Transmission System</u>" shall mean Transferred Facilities and Non-transferred Facilities taken as a whole.

"<u>Transmission Transaction</u>" shall mean Transmission Service scheduled by the Company to the extent permitted by the MISO ITC Agreement.

"<u>Transmission User</u>" shall have the meaning given to the term "Users" in the MISO OATT.

"Underpaid Party" shall have the meaning given in Section 3.3.4(c).

"Underpayment" shall have the meaning given in Section 3.3.4(c).

"Willful Misconduct" shall mean (i) an act or omission by (A) the Company in the performance of its duties or obligations under this Agreement or (B) any Affiliate thereof that provides services to or for the benefit of the Company in the performance of such services, in either case, that is in disregard of a known, reasonably knowable or reasonably obvious risk that harm to the Company, any Transmission Owner, or any of the facilities included in the Transmission System is likely to follow or (ii) a deliberate breach of this Agreement by (A) the Company in respect of any of its duties or obligations hereunder or (B) any Affiliate thereof that provides services to or for the benefit of the Company, in either case, in the performance of such services; provided, however, that an act by the Company in its capacity as an owner of facilities that comprise part of the Transmission System or any act or omission of an Affiliate which owns facilities that are part of the Transmission System in its capacity as such shall not be considered Willful Misconduct.

1.2 Rules of Construction. The following provisions shall be applied wherever appropriate herein:

- 1. "herein," "hereby," "hereunder," "hereof," "hereto" and other equivalent words shall refer to this Agreement as an entirety and not solely to the particular portion of this Agreement in which any such word is used;
- 2. "including" means "including without limitation" and is a term of illustration and not of limitation;
- 3. all definitions set forth herein shall be deemed applicable whether the words defined are used herein in the singular or the plural;
- 4. unless otherwise expressly provided, any term defined in this <u>Article I</u> by reference to any other document shall be deemed to be amended herein to the extent that such term is subsequently amended in such document;
- 5. wherever used herein, any pronoun or pronouns shall be deemed to include both the singular and plural and to cover all genders;
- 6. neither this Agreement nor any other agreement, document or instrument referred to herein or executed and delivered in connection herewith shall be construed against any Person as the principal draftsperson hereof or thereof;
- 7. the Section headings appearing in this Agreement are inserted only as a matter of convenience and in no way define, limit, construe or describe the scope or extent of such section, or in any way affect this Agreement;
- 8. any references herein to a particular Section, Article or Appendix means a Section or Article of, or an Appendix to, this Agreement unless another agreement is specified; and
- 9. the Appendices attached hereto are incorporated herein by reference and shall be considered part of this Agreement.

ARTICLE II AUTHORIZATIONS BY THE TRANSMISSION OWNERS

2.1 Functional Control Authorization. Each Transmission Owner hereby agrees that on and as of the Transmission Service Date the Company shall, and the Company hereby agrees to, (i) assume Functional Control over such Transmission Owner's Transferred Facilities, and (ii) cede to the Midwest ISO those functions set forth in the MISO ITC Agreement to be ceded to the Midwest ISO, in each case for the Initial Term and any Additional Term. The Company shall not exercise direct physical control over the Transferred Facilities except as set forth in this Agreement or in a separate agreement with a Transmission Owner. Notwithstanding the foregoing, the Parties shall not have any rights, duties, or obligations under this Agreement until the Transmission Service Date has occurred, except with respect to the Parties' respective rights, duties, and obligations under Articles V, VI, VII, IX and X hereof, and so much of Article I hereof as is applicable. If a Person becomes a Transmission Owner pursuant to Section 3.4.3, the Company shall assume Functional Control over the Transferred Facilities of such Transmission Owner on the later of (i) the date set forth in the amendment hereto entered into

between the Company and such Transmission Owner pursuant to <u>Section 3.4.3</u> and (ii) the Transmission Service Date.

- 2.2 Non-transferred Facilities. Each Transmission Owner hereby appoints the Company as its agent (and the Company hereby agrees to serve as such agent) to exercise Functional Control over Non-transferred Facilities to the extent necessary for the Company to perform its obligations under this Agreement and the MISO ITC Agreement. Upon the Company's request, each Transmission Owner further agrees to provide the Company with all information relating to its Non-transferred Facilities that is necessary or appropriate for the Company to perform its obligations under this Agreement and the functions set forth in the MISO ITC Agreement. The agency authorization set forth in this Section 2.2 shall not be construed as authorizing the Company to enter into any agreement that creates any liability, costs, or other obligation to be borne by any Transmission Owner that is not expressly set forth in the MISO OATT or to enter into any agreement providing for Interconnection Service with regard to Non-transferred Facilities.
- 2.3 Interconnection Agreements. Each Transmission Owner acknowledges and agrees that the Company is obligated under the MISO ITC Agreement to be responsible for Interconnection Service on the Transferred Facilities. Whether or not the Transmission Owner also is a party to any such Interconnection Agreement, each Transmission Owner authorizes the Company to enter into Interconnection Agreements and related agreements involving its Transferred Facilities (but not its Non-transferred Facilities) in accordance with the MISO ITC Agreement and the Interconnection Procedures. Each Transmission Owner shall, upon the reasonable request of the Company, cooperate in the performance of activities necessary to implement requests for interconnection to its Transferred Facilities and necessary to implement the terms of Interconnection Agreements and other agreements related thereto. To the extent that such Interconnection Agreements require the Company to design, procure, construct, install, and/or maintain upgrades to a Transmission Owner's Transferred Facilities or interconnection facilities between the point of interconnection and a Transmission Owner's Transferred Facilities, such Transmission Owner shall perform such activities (including providing for the design, procurement and construction of the necessary facilities) to the extent directed by, and under the supervision of, the Company. Such Transmission Owner shall timely provide to the Company an estimate of the costs such Transmission Owner expects to incur in the performance of such activities. To the maximum extent permitted under the MISO ITC Agreement and the Interconnection Procedures, the Company shall require an interconnecting Generator or local distribution entity, as the case may be, to provide funds in advance of construction in amounts sufficient to compensate the affected Transmission Owner for the amount estimated by the affected Transmission Owner as the cost to be incurred in connection with such interconnection facilities and any associated interconnection system upgrades and the Company shall remit such funds to the Transmission Owner on a Pass-Through Basis. If the Company is not permitted to require an interconnecting Generator or local distribution entity to provide funding in advance, the Company shall, to the extent permitted by the MISO ITC Agreement and the Interconnection Procedures, assist the Transmission Owners to recover their verifiable costs of performing interconnection activities. In no event shall the Company be responsible for payments to the Transmission Owner other than on a Pass-Through Basis except as follows: (i) to the extent that the Company fails to require advance funds and/or credit support in an amount no less than the amount estimated by the Transmission Owner or as otherwise determined in an appropriate

proceeding as the cost to be incurred in connection with such interconnection facilities and any associated interconnection system upgrades, provided that the Company is permitted to require such amount of advance funds and/or credit support under the MISO ITC Agreement, the Interconnection Procedures and Applicable Laws and Regulations or (ii) if the Company initiates a proceeding (except as is necessary or appropriate to comply with a Commission requirement or Commission policy) to reduce the credit support requirements with respect to interconnection customers generally or that customer in particular from the levels permitted under the Interconnection Procedures without the written consent of the Transmission Owners and such credit approval requirements are reduced as a result of such proceeding. Any interconnection system upgrades and/or interconnection facilities constructed or installed on the Transmission System side of the point of interconnection and such other facilities as may be identified in regard to the interconnection shall become part of such Transmission Owner's Transferred Facilities subject to this Agreement. To the extent the Company possesses property rights in any such interconnection facilities, the Company shall execute appropriate and customary bills of sale, easements, deeds, assignments, and other documents as may be necessary from time to time to convey to such Transmission Owner any property interest that the Company may have in such assets free and clear of all liens and encumbrances created by the Company.

2.4 Network Upgrades. Each Transmission Owner acknowledges and agrees that the Company is obligated under the MISO ITC Agreement to use commercially reasonable efforts to construct Transmission Facilities as directed by the Midwest ISO in order to enable the Midwest ISO to provide Transmission Service pursuant to the MISO OATT. Each Transmission Owner authorizes the Company to enter into agreements with Transmission Customers involving Network Upgrades to its Transferred Facilities which are necessary or desirable to enable the Company to perform its obligations under this Agreement and the MISO ITC Agreement, provided that all such agreements shall be entered into in accordance with the MISO OATT. Each Transmission Owner shall, upon the reasonable request of the Company, cooperate in the performance of activities necessary to fulfill the terms of any agreement involving such Network Upgrades and any agreements related thereto. To the extent that such agreements require the Company to design, procure, construct, install, and/or maintain upgrades or additions to a Transmission Owner's Transferred Facilities, such Transmission Owner shall perform such activities (including providing for the design, procurement and construction of the necessary facilities) to the extent directed by, and under the supervision of, the Company. Transmission Owner shall timely provide to the Company an estimate of the costs such Transmission Owner expects to incur in the performance of such activities. To the maximum extent permitted under the MISO ITC Agreement and the MISO OATT, the Company shall require a Transmission Customer to provide funds in advance of construction in amounts sufficient to compensate the affected Transmission Owner for the amount estimated by the affected Transmission Owner as the cost to be incurred in connection with such Network Upgrades and shall remit such funds to the affected Transmission Owner or Transmission Owners on a Pass-Through Basis. If the Company is not permitted to require a Transmission Customer to provide funding in advance, the Company shall, to the extent permitted by the MISO ITC Agreement and the MISO OATT, assist each affected Transmission Owner to recover its verifiable costs of performing the Network Upgrade activities. In no event shall the Company be responsible for payments to the Transmission Owner other than on a Pass-Through Basis, except as follows: (i) to the extent that the Company fails to require advance funds and/or credit support in an amount no less than the amount estimated by the Transmission Owner or as

otherwise determined in an appropriate proceeding as the cost to be incurred in connection with such Network Upgrades, provided that the Company is permitted to require such amount of advance funds and/or credit support under the MISO ITC Agreement, the MISO OATT and Applicable Laws and Regulations or (ii) if the Company initiates a proceeding (except as is necessary or appropriate to comply with the MISO ITC Agreement, the MISO OATT or a Commission requirement or Commission policy) to reduce the credit support requirements with respect to Transmission Customers generally or that Transmission Customer in particular from the levels permitted under the MISO ITC Agreement and the MISO OATT without the written consent of the Transmission Owners and such credit approval requirements are reduced as a result of such proceeding. Any Network Upgrades constructed or installed on the Transmission System and such other facilities as may be identified shall become part of the applicable Transmission Owner's Transferred Facilities subject to this Agreement. To the extent the Company possesses property rights in any such Network Upgrades, the Company shall execute appropriate and customary bills of sale, easements, deeds, assignments, and other documents as may be necessary from time to time to convey to such Transmission Owner any property interest that the Company may have in such asset free and clear of all liens and encumbrances created by the Company.

2.5 Revenue Collection and Distribution. Prior to the Transmission Service Date the Transmission Owners shall provide to the Company written instructions specifying the manner in which all revenues received for the provision of Transmission Services and other services provided pursuant to the MISO OATT shall be distributed among the Transmission Owners, and the Company shall comply with such instructions unless and until such instructions are revised by the Transmission Owners; provided, however, that all revenues comprising incentive revenues, whether received by the Transmission Owners directly or indirectly through the Company, shall be determined and allocated to the Company and to each Transmission Owner as set forth in Section 4.3.2. To the extent that the proper distribution of revenues requires a Transmission Owner to provide information and/or instructions to the Company, the Company shall be entitled to rely upon such information and/or instructions provided by the Transmission Owner and shall not have any obligation to independently verify such information.

2.6 The Company as Owner of Transmission Facilities; Non-Discrimination.

2.6.1 The Company as Owner of Transmission Facilities. If the Company acquires any facilities included in the Transmission System, it shall operate such facilities as part of an integrated system together with the other facilities included in the Transmission System, but, except as specifically provided in this Agreement, shall not be deemed to be a Transmission Owner pursuant to this Agreement; provided, however, that if an Affiliate of the Company acquires facilities that comprise part of the Transmission System, such Affiliate shall execute an assumption agreement assuming the rights, duties and obligations of the selling Transmission Owner hereunder with respect to such facilities and shall be treated as a Transmission Owner for all purposes under this Agreement. Such Affiliate shall assume the same rights, duties, and obligations under this Agreement as the Transmission Owner from whom such Affiliate purchased the facilities.

2.6.2 **Nondiscrimination.** The Company shall perform its obligations under this Agreement in a manner that does not discriminate in favor of or against the Transferred Facilities of any Transmission Owner or of the Company or its Affiliates, and in furtherance thereof, shall not engage in activities intended to enhance the revenue of the Company or any Transmission Owner relative to another Transmission Owner or the Company. The Company shall treat the Transmission Facilities that are Transferred Facilities but are not owned by the Company and the Transmission Facilities the Company owns as a single integrated business with respect to operations, rate design and other matters affecting the financial return on such assets to their owners and, in that connection, shall not give undue preference to any particular Transmission Facilities. The Company's compliance with this Section 2.6.2 shall be determined taking into consideration the totality of the circumstances, and it shall have an absolute defense to any claim of violation of this Section that the action or inaction complained of was within the authority ceded to the Midwest ISO or undertaken or not undertaken at the direction of the Midwest ISO pursuant to the MISO ITC Agreement. The fact that a Transmission Owner is disadvantaged vis-à-vis any other Transmission Owner or the Company shall not in itself constitute a violation of the Company's obligations under this Section. A Transmission Owner shall not be required to provide evidence of the Company's express intent to disadvantage such Transmission Owner in order to demonstrate that the Company has violated this Section. The Parties agree to submit all disputes relating to this anti-discrimination provision to the Commission for resolution; provided, however, that a Party may institute an Arbitration if the Commission disclaims jurisdiction over a dispute.

ARTICLE III RIGHTS, POWERS, AND OBLIGATIONS OF THE GridAmerica ITC

3.1 **Operation And Planning.**

3.1.1 Standards.

(a) The Company shall participate in the Midwest ISO to the extent set forth in the MISO ITC Agreement. The Company shall perform its obligations under this Agreement, in accordance with Good Business Practice. Notwithstanding the foregoing, in recognition of the fact that the Company has ceded certain functions with respect to the Transmission Facilities to the Midwest ISO pursuant to the MISO ITC Agreement, the Company shall not be deemed to have violated Good Business Practice by entering into the MISO ITC Agreement, any amendments or modifications thereto, or any similar agreements in which one or more Regional Transmission Organizations assumes responsibility for any part of Functional Control of any Transmission Facilities or the operation thereof, and

it shall be absolute defense to any claim that the Company did not adhere to Good Business Practices that (i) the Midwest ISO (or another RTO) had responsibility for the performance of the function in question under the MISO ITC Agreement or any similar agreement or under Applicable Laws and Regulations or (ii) that the Company was acting under instruction of the Midwest ISO (or another RTO) given pursuant to the MISO ITC Agreement or any similar agreement, provided that the Company shall have exercised Good Utility Practice in its actions implementing such instructions.

- **(b)** The Company shall adhere to all applicable reliability guidelines, policies, standards, rules, regulations, orders, license requirements and all other requirements of the Midwest ISO, NERC or the regional reliability council of NERC in which Transferred Facilities are located, each Transmission Owner's specific reliability requirements and operating guidelines applicable on the Transmission Service Date and as thereafter modified by the Transmission Owner with the agreement of the Company to the extent permitted under the MISO ITC Agreement, and all Applicable Laws and Regulations. Disputes regarding a Transmission Owner's specific reliability requirements and operating guidelines shall be submitted to the Midwest ISO for resolution. Pending resolution of such disputes, to the extent permitted under the MISO ITC Agreement, a Transmission Owner's specific reliability requirements and operating guidelines applicable on the Transmission Service Date (or thereafter modified by the Transmission Owner with the consent of the Company) shall be used by the Company with respect to the Transmission Owner's facilities until the issue is resolved.
- **3.1.2 Reliability.** The Company hereby assumes and shall have responsibility for the reliability of the Transferred Facilities to the extent permitted under the MISO ITC Agreement, subject to Applicable Laws and Regulations.
- 3.1.3 Planning and Operating Activities. The Company shall adopt detailed procedures for planning and operating the Transferred Facilities, including procedures for developing plans for the expansion and utilization of the Transferred Facilities and the implementation of the Company's planning and operating responsibilities under the MISO ITC Agreement. The procedures adopted by the Company shall become effective as of the Transmission Service Date for all purposes under this Agreement. If, as of the Transmission Service Date, more than one Transmission Owner disagrees with a procedure adopted by the Company, then those Transmission Owners which so disagree jointly shall have the right to submit the dispute to an independent engineer who is a Non-Market Participant and is reasonably acceptable to the Company; provided,

however, that no such disagreement or dispute resolution process shall delay the occurrence of the Transmission Service Date; and provided further, that the procedures adopted by the Company shall remain in effect pending receipt of the recommendation of the independent engineer. The Company shall adopt and make effective, prospectively, the procedure recommended by the independent engineer to the extent such recommended procedure does not adversely affect the ability of the Company to perform its obligations under the MISO ITC Agreement or to comply with Applicable Laws and Regulations. The planning and operating procedures shall set forth the process by which such procedures may be amended, modified and supplemented from time to time after the Transmission Service Date.

3.1.4 Performance of Regulatory Obligations. The Company shall comply, and shall provide such information to each Transmission Owner as each such Transmission Owner requires to comply, and shall otherwise assist each such Transmission Owner in complying, with existing transmission, reporting, operating, filing, and planning obligations of each such Transmission Owner that are imposed by Applicable Laws and Regulations and which can no longer be performed solely by such Transmission Owner following the Transmission Service Date.

3.2 Other Matters.

- **3.2.1 Pricing.** The Company or one or more Transmission Owners may propose to the Commission such transmission pricing for Transmission Service provided on the Transmission System to the extent permitted under Applicable Laws and Regulations.
- **3.2.2** <u>Ancillary Services</u>. The Company may offer Ancillary Services to the extent permitted by the MISO ITC Agreement, subject to Applicable Laws and Regulations.

3.3 Responsibilities To The Transmission Owner.

- **3.3.1** Relationship. Except to the extent that (a) a Transmission Owner has divested some or all of its Transferred Facilities to the Company or an Affiliate of the Company or (b) the Company and a Transmission Owner have, by separate agreement, agreed to joint ownership of certain Transferred Facilities, the Company shall not by reason of this Agreement, have any ownership interest in the Transferred Facilities or in any revenues or other monies to which a Transmission Owner is entitled hereunder.
- **3.3.2** Avoidance of Damage. If the Company undertakes any action or avoidance of any action in compliance with instructions of the Midwest ISO issued to the Company pursuant to the Midwest ISO's functional

responsibilities under the MISO ITC Agreement, and a Transmission Owner reasonably believes that such action or avoidance of action would cause damage to the Transferred Facilities or any portion thereof or any property of the Transmission Owner affected by the Company's action or avoidance of action, such Transmission Owner shall promptly advise the Company of the expected consequences of the Company's action or inaction and the Company after receipt of such notice shall promptly advise the Midwest ISO of such expected consequences. Notwithstanding the foregoing, any action or avoidance of action by the Company undertaken in compliance with an instruction issued to the Company by the Midwest ISO (after advising the Midwest ISO of the expected consequences) and in accordance with the exercise of Good Business Practice shall not constitute a breach of the Company's obligations under this Agreement.

3.3.3 Duty to Maximize Transmission System Value. The Company shall use commercially reasonable efforts to maximize the long-term value, including net revenues, of the Transmission System so long as such efforts are consistent with its reliability responsibilities, customer service obligations and other obligations under this Agreement, the MISO ITC Agreement and Applicable Laws and Regulations.

3.3.4 Lockbox Account.

(a) The Company shall establish promptly after the execution hereof and shall maintain at all times a lockbox account (together with any subaccounts) (the "Lockbox Account") with an Acceptable Credit Bank selected by the Company (the "Lockbox Bank"). All payments for Transmission Service received by the Company shall be deposited in the Lockbox Account to be allocated among the Company and the Transmission Owners in a manner agreed to by the Parties.

The Company shall:

- (i) establish a collection account (the "Collection Account") which shall be expressly designated as a custodial account established for the joint benefit of the Company and each Transmission Owner into which payments will be made by the Midwest ISO and any other Person making payments to the Company;
- (ii) establish a separate subaccount for each Transmission Owner and the Company to facilitate the distribution of funds to which each Transmission Owner and the Company is entitled on a Pass-Through Basis or as otherwise agreed

- to by the Parties (with respect to each Transmission Owner and the Company, its "Lockbox Subaccount"); and
- (iii) the grant of a security interest to each Transmission Owner in (a) its Lockbox Subaccount and (b) to the extent of its interests therein, the Collection Account and all amounts deposited therein;
- (iv) instruct the Lockbox Bank that funds deposited into the Collection Account may be transferred by the Company only into the Lockbox Subaccounts referred to in Section 3.3.4(a)(ii) pursuant to Section 3.3.4(b).
- (v) instruct the Lockbox Bank that each Transmission Owner and the Company shall have the right, without the consent of any other Person, to withdraw amounts on deposit in such Transmission Owner's or the Company's Lockbox Subaccount as and when desired by such Transmission Owner and/or the Company.
- (b) As soon as practical after receipt of collected funds in the Collection Account, the Company shall cause the Lockbox Bank to transfer the share of such funds belonging to each of the Company and the Transmission Owners to its respective Lockbox Subaccount. The Company and the Transmission Owners acknowledge and agree that the Lockbox Subaccounts and all amounts deposited therein are to constitute the sole and exclusive property of the Transmission Owner or the Company in whose name such Lockbox Subaccount is opened.
- (c) Any Party that acquires actual knowledge that it has received from the Lockbox Bank funds in excess of those to which such Party is entitled hereunder (an "Overpayment") promptly shall provide the Company with written notice of such overpayment. If within one (1) year of receipt of funds by a Party from the Lockbox Bank such Party determines that it has received funds less than those to which such Party is entitled hereunder (an "Underpayment"), such Party promptly shall provide the Company with written notice of such Underpayment. If the Company receives a notice of an Overpayment Underpayment or an pursuant Section 3.3.4(c), the Company immediately shall (i) notify any other Party that may be affected thereby and (ii) review the books and records of the Company relating to the payment(s) in question to ascertain whether an Underpayment and/or Overpayment occurred. Any disputes regarding the existence of an Overpayment or Underpayment and corrective actions to be taken shall be resolved in accordance with Article VI. If an Overpaid Party seeks

to terminate this Agreement for any reason, then such Overpaid Party, shall, as a condition precedent to its termination of this Agreement, pay to the Company, as agent for and for remittance to each Underpaid Party the amount of any Overpayment such Overpaid Party has received.

3.3.5 The Company shall pay any fees of the Lockbox Bank and shall cooperate with each Transmission Owner in effecting such Transmission Owner's rights under this Section 3.3.5. The Company shall also take such further actions, as such Transmission Owner deems reasonably appropriate or advisable to effectuate the purposes of this Section 3.3.5, to enable any Party to exercise and enforce its rights and remedies under this Section 3.3.5, and to perfect, preserve or protect the interest of such Party in the Lockbox Account contemplated by this Section 3.3.5.

3.4 Additional Obligations.

3.4.1 Indemnification by the Company.

(a) The Company shall indemnify each Transmission Owner from all Losses suffered or incurred by such Transmission Owner and arising out of or caused by any failure of the Company to meet its obligation to use Good Business Practices in connection with the performance of the Company's duties and obligations under this Agreement (any such failure being hereinafter referred to as a "Good Business Practice Breach"); provided, however, that, except as hereinafter provided, the Company shall not be liable for Losses from any claim or series of related claims involving any Good Business Practice Breach (i) unless such Losses exceed \$150,000 in the aggregate, and then only to the extent that such Losses exceed \$150,000 or (ii) if and to the extent that such Losses arising from any claim or series of related claims occurring in any calendar year plus all Company Payments with respect to all other claims occurring in such calendar year are greater than the Liability Cap Amount for such calendar year. Notwithstanding the foregoing, none of the limitations on liability contained in this Section 3.4.1(a) shall apply in respect of any Losses arising out of Gross Negligence or Willful Misconduct (and the Company shall be fully liable for any breach of any provision of this Agreement arising out of or caused by Gross Negligence or Willful Misconduct or breaches by the Company of its obligations under Sections 10.7 and 10.14). To the extent that a claim asserted against the Company relates to Losses suffered by more than one Transmission Owner, then the Transmission Owner that asserted such claim and such other Transmission Owners shall share the indemnification payments made by the Company in respect thereof in proportion to the Losses suffered by each.

- (b) The term "Company Payments" means, with respect to any calendar year, the sum of the following determined as of the time in question:
 - (i) the aggregate amount of all indemnification payments actually made by the Company under Section 3.4.1(a) with respect to claims occurring in such calendar year with respect to Good Business Practice Breaches (excluding payments made out of insurance proceeds); plus
 - (ii) the aggregate amount of all Losses actually paid or due and payable by the Company (excluding payments made out of insurance proceeds and excluding the application of any indemnification payments from the Transmission Owners pursuant to Section 4.2.4(c)) with respect to Third Party Claims (other than claims by the Transmission Owners pursuant to Section 3.4.1(a) of this Agreement) occurring in such calendar year to the extent such claims involve Good Business Practice Breaches; plus
 - (iii) the aggregate amount of all "Managing Member Payments" referred to in Section 11.8(f)(i) of the LLC Agreement that are paid by the Managing Member in respect of claims occurring in such year;

provided, however, that no amount paid by the Company as a result of Gross Negligence or Willful Misconduct shall constitute a Company Payment. For purposes of determining the amount of the Company Payments in respect of any calendar year, a claim shall be deemed to have occurred in such calendar year if the facts, circumstances or events which first gave rise to a Loss occurred during such calendar year, regardless of when the claim was asserted or when any particular element of Loss was incurred.

- (c) For the avoidance of doubt, the Company shall have no indemnification obligation under Section 3.4.1(a) with respect to Losses arising out of any claim occurring in any calendar year and which arises out of a Good Business Practice Breach to the extent the amount of such Losses in respect of such claim plus all Company Payments with respect to all other claims in such calendar year are greater than the Liability Cap Amount for such calendar year.
- (d) Any Transmission Owner asserting a claim for damages against the Company shall, promptly after the initiation of such claim, give notice thereof to the other Transmission Owners and the Members, which notice must include a reasonably detailed description of the

basis for such claim. The Company and the Transmission Owners agree that if any Member asserts a claim against the Managing Member arising out of the same facts and circumstances that give rise to the claim by the Transmission Owner or Transmission owners asserting a claim against the Company pursuant to Section 3.4.1(a), such claim may, at the written request of such Member received by the Managing Member within thirty (30) days of the date on which such Member received notice of the initiation of such claim, be consolidated with, and determined in the same proceeding as the claim for indemnity asserted against the Company pursuant to Section 3.4.1(a).

- **3.4.2** Inspection and Auditing Procedures. The Company hereby grants to each Transmission Owner and its outside auditors and consultants such access to the Company's books and records as is necessary to verify and audit compliance by the Company with the requirements of this Agreement. Such access shall be at reasonable times and under reasonable conditions. The Transmission Owners shall use reasonable efforts to conduct any such audits in concert but shall not be prohibited from exercising their rights under this Section 3.4.2 individually. The Company shall also comply with the accounting and reporting requirements of Governmental Authorities having jurisdiction over the Company with respect to the business aspects of its business operations.
- 3.4.3 Agreements with Additional Transmission Owners. The Company may, from time to time, in its discretion, enter into an amendment to this Agreement for the purpose of adding as a Party hereto any Person, including a Person that is an Affiliate of the Company, that has Transmission Facilities that it proposes to subject to this Agreement upon a determination by the Company, in its reasonable discretion, that subjecting such Transmission Facilities to this Agreement (x) will not result in any significant detriment to existing Transmission Owners in their capacity as such and (y) is likely to result in long term net benefits to the Company; provided, however, that either (i) the terms of such amendment must not be materially more favorable to the counterparty thereto than the terms applicable to the other Transmission Owners contained herein or (ii) the Company must offer to make any such more favorable terms available to all Transmission Owners on a nondiscriminatory basis; provided, further, that, so long as it does not significantly adversely affect any other Transmission Owner, such an amendment to this Agreement that is with an owner of Transmission Facilities that is a Non-Market Participant may be made on terms that are different than those set forth herein. The Company may not exercise Functional Control over any Transmission Facilities (other than those it owns) except pursuant to this Agreement.

- **3.4.4 Insurance.** At all times during the effectiveness of this Agreement, the Company shall maintain insurance of the types and in the amounts agreed to by the Company and the Transmission Owners. If a Transmission Owner requests the Company to obtain insurance in addition to the types or amount of coverage agreed to, the Company shall obtain such insurance, provided that such Transmission Owner shall pay all of the costs thereof. The insurers or reinsurers and the Company, to the extent that it has a right of subrogation, shall waive all rights of subrogation against the Transmission Owners.
- 3.4.5 <u>Coordination with State Securitization Obligations</u>. The Parties acknowledge that a portion of the revenues payable under the MISO OATT in respect of Transmission Service provided over a particular Transmission Owner's Transferred Facilities may be securitized, pledged, or otherwise subject to superior rights of third parties ("securitized"). The Company shall cooperate with such Transmission Owner with respect to such securitization obligations.
- 3.5 <u>Information</u>. Subject to the confidentiality provisions set forth in <u>Section 10.14</u>, the Company shall provide such information to Transmission Owners as is necessary or appropriate for the Transmission Owners to perform their obligations under this Agreement.

ARTICLE IV RIGHTS, POWERS, AND OBLIGATIONS OF THE TRANSMISSION OWNERS

4.1 Operation and Planning.

4.1.1 Standards.

- (a) Each Transmission Owner shall physically operate its Transferred Facilities in accordance with this Agreement and the MISO OATT and shall comply with the procedures, manuals, and directions of the Company issued in compliance with this Agreement and the MISO ITC Agreement. The Transmission Owners shall provide the services to be performed individually by Transmission Owners as set forth on Schedule 5A attached hereto as Appendix D. No Transmission Owner shall take any action that intentionally interferes with the performance of any function by the Company other than to the extent necessary to avoid an Emergency or during an Emergency. This Section 4.1.1 (a) shall apply to the Company if the Company owns facilities that comprise part of the Transmission System.
- (b) Each Transmission Owner shall operate its Transferred Facilities in accordance with Good Utility Practice, and shall adhere to all applicable reliability guidelines, policies, standards, rules,

regulations, orders, licensing requirements and requirements of NERC or the regional reliability council of NERC in which a facility owned by it is located. If a Transmission Owner believes that a direction given by the Company is in contravention of the requirements of NERC or a regional reliability council of NERC, it shall immediately advise the Company. If a Transmission Owner and the Company disagree as to whether such direction is in contravention of the requirements of NERC or a regional reliability council of NERC, the Company shall submit such dispute to the Midwest ISO for resolution. No Transmission Owner shall engage in behavior which manipulates available transfer capability to the detriment of Transmission Users. The Transmission Owners shall perform all duties and functions specified for the Transmission Owners in this Agreement. This Section 4.1.1(b) shall apply to the Company if the Company or any of its Affiliates becomes an owner of facilities that comprise part of the Transmission System.

4.1.2 Transmission Maintenance.

- (a) Unless otherwise mutually and expressly agreed to by the Company and a Transmission Owner, each Transmission Owner (including the Company if it owns Transmission Facilities that comprise part of the Transmission System) shall repair, maintain, and replace its Transferred Facilities consistent with Good Utility Practice; provided, however, that no Transmission Owner shall exercise Functional Control over the Transferred Facilities. Except as may be required to exercise Functional Control, the Company shall have no obligation or any right to repair, maintain, or replace a Transmission Owner's Transferred Facilities or Non-transferred Facilities.
- (b) The Transmission Owner shall obtain the Company's approval for all planned maintenance of such Transmission Owner's Transferred Facilities. Each Transmission Owner shall submit its planned maintenance schedules to the Company. All proposed planned maintenance schedules shall be evaluated by the Company on a non-discriminatory basis. The Company will coordinate with both the operations and planning personnel of the Transmission Owners for analysis and planning purposes when a transmission maintenance request is received. If requested by the Company, a Transmission Owner will provide to the Company an estimate of the costs associated with potential changes to an approved maintenance schedule or any part thereof.
- (c) After receiving a planned maintenance request for non-critical Transmission Facilities (as such term is used in the MISO ITC Agreement), the Company shall, in a timely manner, either

approve the request or deny the request and provide an alternative time frame in which the maintenance can be performed; provided that any such approval shall not be final and shall be subject to further modification after receipt by the Company of the Midwest ISO's approval of the schedule for critical Transmission Facilities (as such term is used in the MISO ITC Agreement). The Company shall determine in a nondiscriminatory manner and with the use of appropriate analytical detail whether, and, if so, the extent to which, a planned transmission maintenance request for non-critical Transmission Facilities affects available transfer capability, Ancillary Services, the security of the Transmission System, and any other relevant matters. The Company shall submit the maintenance schedule for critical Transmission Facilities to the Midwest ISO for approval in accordance with the MISO ITC Agreement. The Company shall promptly notify the Transmission Owner of the Midwest ISO's approval, disapproval or modification and shall modify the schedule for maintenance on non-critical Transmission Facilities if appropriate. The Company shall communicate the final planned maintenance schedule for noncritical and critical Transmission Facilities to the appropriate Transmission Owner in a timely manner.

- (d) If the Company revokes a planned transmission maintenance outage schedule for non-critical Transmission Facilities of a Transmission Owner after it has become final, the Company shall notify such Transmission Owner of the decision to revoke approval of the maintenance schedule as soon as possible after the circumstances arise that create the need for the revocation. If such Transmission Owner incurs any additional costs associated with the deferred transmission maintenance, the Company shall compensate on a Pass-Through Basis such Transmission Owner for all such verifiable costs collected by the Midwest ISO in rates or other charges. Revocation of previously approved maintenance schedules and compensation for related costs shall be applied in a non-discriminatory basis.
- (e) The Company shall document all planned transmission maintenance requests, the disposition of those requests and all data supporting the disposition of each request.
- (f) A Transmission Owner shall notify the Company when such Transmission Owner is performing maintenance on a facility that could reasonably be expected to result in unplanned outages within the Transmission System. The Company shall coordinate with the Transmission Owners to implement unplanned transmission maintenance as the need may arise. In an Emergency, the applicable Transmission Owner shall immediately notify the

Company of any necessary Emergency transmission maintenance and the Transmission Owner shall document all events of unplanned transmission maintenance and all data related to the Emergency necessitating such maintenance. Prior approval by the Company for such Emergency transmission maintenance will not be required.

4.1.3 Construction of New Facilities.

- The Transmission Owners acknowledge and agree that the (a) Company may require a Transmission Owner to construct planned Transmission Facilities, whether the construction is required pursuant to a direction by the Midwest ISO to the Company or at the initiative of the Company to the extent permitted by the MISO ITC Agreement. Each Transmission Owner and the Company if it is an owner of facilities which comprise part of the Transmission System shall construct at its sole cost and expense new Transmission Facilities reviewed, approved, and ordered to be built by the Company in accordance with planning processes and protocols established by the Company and the Midwest ISO pursuant to the MISO ITC Agreement and the planning procedures adopted by the Company. The Company will develop non-discriminatory criteria consistent with this Section 4.1.3 to determine which Party (including the Company if it is an owner of facilities comprising part of the Transmission System) will be obligated to construct the new facilities. If the new Transmission Facilities will be directly connected to the existing facilities of one Transmission Owner or the Company if it is an owner of facilities comprising part of the Transmission System, that Transmission Owner or the Company will be obligated to construct the new facilities if required by the Company. If two or more Transmission Owners or the Company as an owner of facilities comprising the Transmission System will be interconnected directly to the new facilities, the Company will assign construction responsibilities in accordance with the non-discriminatory criteria stated in the planning procedures adopted by the Company. Prior to the acceptance by the Commission of coordinated planning processes and protocols developed by the Midwest ISO and the Company, the planning and construction of new facilities shall be governed by the planning processes and protocols of the Midwest ISO and the planning procedures adopted by the Company, to the extent such procedures are not inconsistent with the planning processes and protocols of the Midwest ISO.
- (b) The non-discriminatory criteria shall include cost allocation methods for achieving equity among affected Transmission Owners and the Company in circumstances where a

disproportionate, but otherwise unavoidable, burden is placed on one or more Transmission Owners or the Company for the construction of new Transmission Facilities. Any Transmission Owner may propose its own cost allocation methodology to the Commission for the purpose of achieving equity in particular instances.

- (c) If a Transmission Owner that is obligated to construct new Transmission Facilities requests that the Company construct such facilities, the Company shall have the right but not the obligation to do so. If the Company declines to construct such facilities, the Transmission Owner initially obligated to construct the facilities shall remain obligated to do so.
- (d) A Transmission Owner, or the Company as an owner of facilities comprising part of the Transmission System, may satisfy its construction responsibilities by arranging that another party will construct, finance, and/or own the new Transmission Facilities. If the third party fails to perform within the period of time specified by the Company for construction of such new facilities, the Transmission Owner or the Company, in the case of facilities for which the Company has accepted the obligation to construct, shall remain obligated to undertake the construction. The Company shall determine in a non-discriminatory manner whether such third party can meet all necessary criteria for financing and/or owning Transmission Facilities interconnected to the Transmission System. A determination by the Company that such third party fails to meet all necessary criteria shall not relieve the Transmission Owner of the obligation to construct the facilities. Any such third party owning Transmission Facilities shall be required to become a Party to this Agreement pursuant to Section 3.4.3.
- (e) The cost of new Transmission Facilities constructed by Transmission Owners pursuant to this <u>Section 4.1.3</u> shall be recovered in accordance with the MISO OATT and the revenues associated with such cost recovery received by the Company, shall be distributed in accordance with this Agreement.
- **4.1.4** Acquisition. If a Transmission Owner acquires Transmission Facilities that are not part of the Transmission System, such facilities shall become subject to this Agreement provided that (i) such Transmission Facilities are physically interconnected with any of the Transmission Facilities owned or Functionally Controlled by the Company or, if such Transmission Facilities are not physically interconnected with any of the Transmission Facilities owned or Functionally Controlled by the Company, the Company determines, in its reasonable discretion, that

subjecting such Transmission Facilities to this Agreement will result in net benefits to the Company, (ii) such Transmission Owner consents and (iii) such action does not violate Applicable Laws and Regulations. If Transmission Facilities are added to the Transmission System as set forth above, the Company and the Transmission Owner transferring the facilities shall amend such Transmission Owner's subappendix to Appendix A to include such Transmission Facilities.

4.1.5 Performance of Regulatory Obligations. Each Transmission Owner and the Company if it owns facilities that comprise part of the Transmission System shall comply, and shall provide such information to the Company as it requires to comply, and shall otherwise assist the Company in complying, with transmission, reporting, operating, filing, and planning obligations of the Company that are imposed by Applicable Laws and Regulations.

4.2 Additional Obligations.

4.2.1 <u>Information.</u> Subject to the confidentiality provisions set forth in Section 10.14, each Transmission Owner shall provide such information concerning its Transferred Facilities to the Company as is necessary or appropriate for the Company to perform its obligations under this Agreement, the MISO OATT and the MISO ITC Agreement.

4.2.2 Facilities Access.

- (a) Each Transmission Owner shall upon reasonable notice allow the Company access to its Transferred Facilities as is necessary (i) to verify and audit compliance by such Transmission Owner with this Agreement or (ii) for the Company to perform its obligations under this Agreement and the MISO ITC Agreement. Such access shall be at reasonable times and under reasonable conditions. Representatives of the Company shall comply with a Transmission Owner's safety regulations when accessing such Transmission Owner's Transferred Facilities.
- (b) Each Transmission Owner's rights in its Transferred Facilities shall be subject to the Company's Functional Control of the Transmission System in accordance with the terms of this Agreement and the MISO ITC Agreement. Nothing in this Agreement shall be deemed to restrict or to prohibit access to Transferred Facilities by the Transmission' Owner that owns such Transferred Facilities, or those acting under its authority.
- **4.2.3** <u>Inspection and Auditing Procedures.</u> Each Transmission Owner shall grant the Company and its outside auditors and consultants such access to such Transmission Owner's books and records as is necessary (i) to verify

and audit compliance by such Transmission Owner with this Agreement or (ii) for the Company to perform its obligations under this Agreement and the MISO ITC Agreement. Such access shall be at reasonable times and under reasonable conditions. Each Transmission Owner shall also comply with the accounting and reporting requirements of any Governmental Authorities having jurisdiction over the Transmission Owner with respect to aspects of the Company's business operations. Contacts between officers, employees, and agents of the Company and those of a Transmission Owner, pursuant to this Section 4.2.3 shall be strictly limited to the purpose of this Section 4.2.3.

4.2.4 <u>Indemnification by Transmission Owners.</u>

- (a) Each Transmission Owner ("Indemnifying Owner") shall indemnify and hold harmless each other Transmission Owner ("Indemnified Owners") from any Losses arising from the Indemnifying Owner's performance, neglect or breach of its obligations under this Agreement (whether arising from a finding of negligence, strict liability or other fault or responsibility), except in cases where, and only to the extent that, the negligence or willful misconduct of any other Indemnified Owner contributed to the Loss. In the event two or more Transmission Owners have an indemnification obligation under this Section 4.2.4(a) with respect to the same matter, such indemnification obligation shall be borne by such indemnifying Transmission Owners in proportion to each such indemnifying Transmission Owner's comparative fault. If the Company acquires ownership of facilities that comprise part of the Transmission System, the Company shall be treated as if it were a Transmission Owner for purposes of this Section 4.2.4(a). Notwithstanding any other provision of this Agreement, no Transmission Owner shall be liable to any other Transmission Owner for any actions taken in accordance with the instructions or direction of the Company or the Midwest ISO except in cases of the negligence or intentional wrong-doing of the first Transmission Owner.
- (b) Except with respect to its Transferred Facilities or as otherwise provided in the MISO OATT or this Agreement, no Transmission Owner shall be liable for any costs or expenses relating to the operation, repair, maintenance, or improvement of any of the Transmission Facilities constituting part of the Transmission System that are owned, operated, or controlled by any other Transmission Owner or the Company.
- (c) Each Transmission Owner shall severally and not jointly, on a pro rata basis as described below, indemnify and hold harmless the Company from any Losses (determined after application of the

proceeds of any applicable insurance policies) suffered or incurred by the Company (but excluding any lost profits and similar damages suffered or incurred by the Company) resulting from any claim by a third party by reason of the Company's acts, omissions or alleged acts or omissions arising under this Agreement, the MISO ITC Agreement or the MISO OATT or by reason of any of the Midwest ISO's acts, omissions or alleged acts or omissions ("Third Party Claims"); provided, however, that such indemnity does not cover (i) Losses arising out of any action or inaction (A) of the Company taken in bad faith, (B) of the Company that constitutes Gross Negligence or Willful Misconduct or (C) of the Company in its capacity as an owner of facilities that comprise part of the Transmission System, (ii) Losses arising out of Third Party Claims occurring in any calendar year which involve any Good Business Practice Breach, to the extent the Losses paid or due and payable by the Company in respect of such Third Party Claim plus all Company Payments with respect to all other claims occurring in such calendar year are less than the Liability Cap Amount for such calendar year or (iii) Losses from any claim or series of related claims unless such Losses exceed \$150,000 and then only to the extent that such Losses exceed \$150,000. The pro rata liability of each Transmission Owner under this Section 4.2.4(c) shall be equal to a percentage of the aggregate amount of the Losses equal to the percentage that such Transmission Owner's Net Plant bears to the aggregate Net Plant as of the time in which the claim giving rise to the Losses in question occurred. The obligations of each Transmission Owner set forth in this Section 4.2.4(c) shall terminate automatically for all purposes with respect to all Losses from claims arising after the Initial Member ceases to be the Managing Member for any reason. Notwithstanding anything to the contrary in this Agreement, the Transmission Owners' indemnification obligations under this Section 4.2.4(c) shall terminate automatically for all purposes in the event one or more Transmission Owners and/or former Transmission Owners pay, in the aggregate, an amount equal to \$25,000,000 in any year pursuant to this Section 4.2.4(c) (the "Indemnity Cap"), regardless of when the claims which gave rise to the indemnification obligations occurred and regardless of whether the claims involve a Good Business Practice Breach ("Payment Event"); provided, however, if prior to the occurrence of the Payment Event, if the Company owns Transmission Facilities having a Fair Market Value equal to or greater than \$250,000,000 in the aggregate at the time of the acquisition, the indemnification obligations of the Transmission Owners under this Section 4.2.4(c) shall not terminate upon the occurrence of a Payment Event, but nothing in this proviso is a limitation on the automatic termination of the

Transmission Owners' indemnification obligations as provided above in the event the Initial Member ceases to be the Managing Member for any reason.

(d) For avoidance of doubt, no indemnification obligation is owed by the Transmission Owners under Section 4.2.4(c) with respect to Losses arising out of any claim occurring in any calendar year and which arises out of a Good Business Practice Breach to the extent that the amount of such Losses plus all Company Payments with respect to all other claims occurring in such calendar year are less than the Liability Cap Amount for such year.

4.2.5 Administration of Third Party Claims.

- Promptly after receiving notice that a third party has commenced a (a) claim that would be subject to the indemnification provisions of either Section 3.4.1 or 4.2.4, each Party entitled to indemnification (individually and collectively, "Indemnified Party") under such Sections shall in turn give written notice of that claim to each Party obligated provide the indemnification Section (individually and collectively, "Indemnifying Party"). The written notice shall include reasonable detail in light of the circumstances then known to the Indemnified Party. Failure to give such notice, or any notice, will not relieve the Indemnifying Party from its obligations under such Sections except where, and then solely to the extent that, such failure actually and materially prejudices the rights of the Indemnifying Party.
- **(b)** After receiving the written notice described in Section 4.2.5(a), the Indemnifying Party will have the right but not the duty to defend such claim; provided, however, that the Indemnifying Party shall acknowledge in writing its indemnification obligations hereunder with respect to such claim. If, with respect to a claim, the Indemnified Party will retain liability for a material amount in connection with such claim, the Indemnifying Party shall not have the right to assume the defense of such claim hereunder. This right to defend will include the right to pursue any strategy in defending such claim, file any and all pleadings, pursue discovery in any manner, make whatever arguments the Indemnifying Party deems appropriate, and to select counsel of its choice to defend the claim, provided that (i) the Indemnified Party gives its written consent to counsel selected by the Indemnifying Party (which consent shall not be unreasonably withheld) and (ii) the Indemnifying Party conducts the defense of such claim actively and diligently. Attorneys' fees incurred by counsel selected by the Indemnifying Party in defending the claim, as well as other costs and expenses associated with defending the claim, will be paid by the

Indemnifying Party. If the Indemnifying Party consists of one or more of the Transmission Owners, the decision to defend a claim and the administration of such claim as provided in this Section 4.2.5(a) shall be made by Transmission Owners whose collective Net Plant exceeds 50% of the aggregate Net Plant of all Transmission Owners comprising the Indemnifying Party (the "Majority of Indemnifying Transmission Owners"). Each Transmission Owner agrees that the Majority of Indemnifying Transmission Owners shall have no liability to any other Transmission Owner in connection with the administration of claims under this Section 4.2.5 except to the extent the Majority of Indemnifying Transmission Owners exercise their authority hereunder in bad faith or in a grossly negligent manner.

- (c) If the Indemnifying Party assumes the defense of such claim, the Indemnified Party agrees to reasonably cooperate in such defense as long as the Indemnified Party is not materially prejudiced thereby. As long as the Indemnifying Party is conducting the defense of such claim actively and diligently, the Indemnified Party may retain separate co-counsel at its own cost and expense and may participate in the defense of such claim, though ultimate control of the claim's defense shall remain with the Indemnifying Party.
- (d) In the event the Indemnifying Party does not assume the defense of such claim, or ceases to conduct the defense of such claim actively and diligently, or has a conflict of interest with the Indemnified Party, (i) the Indemnified Party may defend against, and, with the prior written consent of the Indemnifying Party (which consent shall not be unreasonably withheld), consent to the entry of any judgment or enter into any settlement with respect to, such claim, (ii) the Indemnifying Party will reimburse the Indemnified Party promptly and periodically for the costs of defending against such claim, including reasonable attorneys' fees and expenses, and (iii) the Indemnifying Party will remain responsible for any Losses the Indemnified Party may suffer as a result of such claim to the full extent provided in Sections 3.4.1 and 4.2.4.
- (e) If after the making of any indemnification payment the amount of Losses to which such payment relates is reduced by recovery, settlement or otherwise under any insurance coverage, or pursuant to any claim, recovery, settlement or payment by or against any other Person, the amount of such reduction (less any costs, expenses, premiums or Taxes incurred in connection therewith) will promptly be repaid by the Indemnified Party to the Indemnifying Party. Upon making any indemnification payment, the Indemnifying Party will, to the extent of such indemnification

payment, be subrogated to all rights of the Indemnified Party against any third party in respect of the Losses to which the indemnification payment relates; provided that (i) the Indemnifying Party is then in compliance with its obligations under this Agreement in respect of such Losses, and (ii) until the Indemnified Party recovers full payment of such Losses, all claims of the Indemnifying Party against any such third party on account of said indemnification payment will be subrogated and subordinated in right of payment to the Indemnified Party's rights against such third party. Without limiting the generality or effect of any other provision of this Section 4.2.5(e), each such Indemnified Party and Indemnifying Party will duly execute upon request all instruments reasonably necessary to evidence and perfect the above-described subrogation and subordination rights.

4.3 **Payments to the Company**

- Management Fee. Each Transmission Owner shall severally and not 4.3.1 jointly, pay the Company its pro rata share of the annual Management Fee. Each Transmission Owner's pro rata share of the Management Fee in each year shall be equal to the percentage of the Management Fee equal to the percentage that such Transmission Owner's Net Plant bears to the aggregate Net Plant of all Transmission Facilities subject to the Functional Control of the Company pursuant to this Agreement in such year. The amount of the Initial Management Fee, as adjusted pursuant to Section 6.3(b) of the LLC Agreement, may be increased upon and in connection with the addition of a Transmission Owner as a Party to this Agreement pursuant to Section 3.4.3 of this Agreement by an amount agreed upon between the Company and such additional Transmission Owner; provided, however, that (a) no such increase in the Initial Management Fee shall result in an increase in the aggregate amount of the Initial Management Fee payable by the Original GridAmerica Companies and (b) after giving effect to such increase in the Initial Management Fee, the several obligation of each Transmission Owner to pay its pro rata share of the Initial Management Fee under this Section 4.3 shall not exceed a percentage of the increased Initial Management Fee equal to the percentage that such Transmission Owner's Net Plant bears to the aggregate Net Plant of all Transmission Facilities subject to the Functional Control of the Company pursuant to this Agreement in such year, adjusted to include the Net Plant of the additional Transmission Owner, as of the effective date of such increase in the Initial Management Fee.
- **4.3.2** <u>Incentive Compensation</u>. In addition to the Management Fee payable to the Company, each Transmission Owner shall pay to the Company incentive compensation pursuant to such incentive compensation arrangements as are agreed from time to time between the Company and such other Transmission Owners as shall agree thereto. The Company

shall, from time to time but no less frequently than within thirty (30) days after the Transmission Service Date and each anniversary thereof, propose to the Transmission Owners incentive compensation arrangements designed to encourage the efficient and enhanced operation of the Transferred Facilities without regard to any benefit or detriment to other businesses and activities of the Transmission Owners, including their electric generation businesses and activities, which may result from implementation of such proposals. Any incentive arrangements between the Company and the Transmission Owners shall provide that (a) not less than 25 percent of net incentive revenues which are earned through the implementation of Company's proposals, or otherwise as a result of the Company's exercise of Functional Control over the Transferred Facilities, shall be payable to the Company as incentive compensation; and (b) each Transmission Owner shall receive a percentage of the balance of such net incentive revenues equal to the percentage that such Transmission Owner's Net Plant bears to the aggregate Net Plant of all Transmission Facilities subject to the Functional Control of the Company pursuant to this Agreement in such year.

4.4 Transmission Owners' Remedies.

- 4.4.1 Accounting Failures. If an Accounting Failure occurs, at the written request of a majority of the Transmission Owners (each Transmission Owner having one vote), the Company shall retain and pay for the services of an accounting firm of national reputation, which accounting firm shall conduct an examination and analysis of the accounting procedures of the Company and shall be required to issue a written report regarding such examination and analysis, including a description of any material weaknesses in the Company's accounting procedures discovered by such accounting firm, and recommendations to correct such weaknesses. The Company shall provide to the Transmission Owners copies of all draft and final reports prepared by such accounting firm and within thirty (30) days after issuance of a report, the Company shall provide a response thereto delivered to the Transmission Owner which states the extent to which the Company accepts the report's findings and the basis for any objections.
- 4.4.2 Operational Failures. If an Operational Failure occurs, at the written request of a majority of the Transmission Owners (each Transmission Owner having one vote), the Company shall retain and pay for the services of a consulting firm of national reputation with experience in electric energy operations and management and which is a Non-Market Participant. Such consulting firm shall conduct an examination and assessment of the Company's operating procedures, management, staffing and such other matters as such consulting firm deems appropriate under the circumstances considering the nature of the Operational Failure and shall be required to issue a written report regarding such examination and assessment. The Company shall provide to the Transmission Owners

copies of all draft and final reports prepared by such consulting firm and within thirty (30) days after issuance of a report, the Company shall provide a response thereto delivered to the Transmission Owners which states the extent to which the Company accepts the report's findings and the basis for any objections.

- 4.4.3 Removal of Managing Member for Cause. If a Super Majority of Transmission Owners shall have delivered written notice that they have elected to remove the Managing Member for Cause, the Company shall cause the Managing Member to be removed in accordance with the provisions of the LLC Agreement; provided, however, that (x) if the Managing Member contests its removal pursuant to the terms of the LLC Agreement, the Company shall acknowledge and shall cause the Managing Member to acknowledge the rights of the Transmission Owners to participate in any proceeding to resolve the dispute over the removal pursuant to Section 10.3 of the LLC Agreement or Section 6.3 and (y) no removal of the Managing Member shall be effective unless and until approved by the Commission.
- **4.4.4** Replacement of Managing Member. If a Managing Member is removed as a result of an Early Termination Event under the LLC Agreement, including a removal for Cause, the Company shall cause the successor Managing Member to be selected by vote of a Super Majority of Transmission Owners.

ARTICLE V TERM AND TERMINATION

5.1 **Term.** This Agreement shall become effective as to the Company and each Transmission Owner on the Effective Date or the date set forth in the amendment hereto entered into between the Company and a Transmission Owner pursuant to Section 3.4.3. For each Transmission Owner and the Company, this Agreement shall continue in effect until the end of the Initial Term. Notwithstanding the scheduled termination hereof at the end of the Initial Term, this Agreement automatically shall be extended as to a Transmission Owner for an additional term of two (2) years at the end of the Initial Term, unless written notice of termination hereof is given by such Transmission Owner to each other Party at least six months prior to the last day of the Initial Term; provided, however, that, if at the time such notice of termination is given by such Transmission Owner, applicable provisions in the Midwest ISO Transmission Owners Agreement governing the right of a "Transmission Owner" thereunder to withdraw from the Midwest ISO specify a longer minimum time for notice of withdrawal, then, unless the Commission shall otherwise approve, upon the effectiveness of the termination of this Agreement as to any Transmission Owner, such Transmission Owner shall automatically be and become a member of Midwest ISO for a term of not less than the

minimum notice period for withdrawal specified in the Midwest ISO Transmission Owners Agreement (measured from the date of the notice of termination delivered hereunder) less six months. Following any such additional term, the term hereof automatically shall be extended as to a Transmission Owner for successive additional terms of two (2) years each (any additional term, whether following the Initial Term or an additional term, being an "Additional Term") unless written notice of termination hereof is given by such Transmission Owner to each other Party at least six months prior to the last day of the then existing Additional Term; provided, however, that, if at the time such notice of termination is given by such Transmission Owner, applicable provisions in the Midwest ISO Transmission Owners Agreement governing the right of a "Transmission Owner" thereunder to withdraw from the Midwest ISO specify a longer minimum time for notice of withdrawal from the Midwest ISO, then, unless the Commission shall otherwise approve, upon the effectiveness of the termination of this Agreement as to any Transmission Owner, such Transmission Owner shall automatically be and become a member of Midwest ISO for a term of not less than the minimum notice period for withdrawal specified in the Midwest ISO Transmission Owners Agreement (measured from the date of the notice of termination delivered hereunder) less six months.

Each Transmission Owner may withdraw from the GridAmerica ITC and this Agreement shall terminate with respect to such Party under the following circumstances: (a) if a Transmission Owner delivers notice of its intent to contribute its Transferred Facilities to the Company pursuant to Section 5.1(a) of the Master Agreement prior to the date thirty (30) months after the Effective Date, then each Transmission Owner may withdraw from the GridAmerica ITC and terminate its participation under this Agreement by providing all of the other Parties written notice of such withdrawal within thirty (30) days of receipt of the notice of such contribution; or (b) if no Transmission Owner has delivered notice of its intent to contribute its Transferred Facilities to the Company pursuant to Section 5.1(a) of the Master Agreement prior to the expiration of the twenty-ninth (29th) month following the Effective Date, then each Transmission Owner may withdraw from the GridAmerica ITC and terminate its participation under this Agreement by providing all of the other Parties written notice of such withdrawal within the thirtieth (30th) month following the Effective Date. Any withdrawal from the GridAmerica ITC and termination of participation under this Agreement shall be effective on the first day of the seventh month following the month in which notice thereof is delivered in accordance with Section 10.9; provided, however, that, notwithstanding the foregoing, unless the Commission shall otherwise approve, no withdrawal by any Transmission Owner from the GridAmerica ITC pursuant to Section 5.1(a) or Section 5.1(b) shall be or become effective unless and until such Transmission

Owner becomes a member of the Midwest ISO. Upon receipt of any notice of withdrawal from the GridAmerica ITC and termination of participation under this Agreement pursuant to Section 5.1(a) or Section 5.1(b), each other Transmission Owners shall have the right, exercisable within thirty (30) days of receipt of such notice, to withdraw from the GridAmerica ITC and terminate its participation under this Agreement; provided, however, that, notwithstanding the foregoing, unless the Commission shall otherwise approve, no withdrawal by any Transmission Owner from the GridAmerica ITC pursuant to this sentence of this Section 5.1 shall be or become effective unless and until such Transmission Owner becomes a member of the Midwest ISO. Without the prior written consent of NGUSA pursuant to Section 5.7 of the Master Agreement, no part of the Transferred Facilities of any Transmission Owner that withdraws from the GridAmerica ITC and terminates its participation under this Agreement shall be included in or managed by an ITC that exercises functions similar in scope to the function exercised by the Company with respect to the Transferred Facilities (determined after taking into account the functions exercised by the Midwest ISO under the MISO ITC Agreement) for a period of one (1) year after the effective date of such Transmission Owner's withdrawal, provided, however, that the foregoing prohibition shall not apply if the Initial Member exercised its resignation rights pursuant to Section 5.7(a) or Section 5.7(b) of the Master Agreement prior to the date of any such withdrawal.

5.2 Termination.

- **5.2.1 No Termination.** No Party shall have any right to terminate or withdraw from this Agreement, nor shall any Transmission Owner have the right to withdraw any of its Transferred Facilities from GridAmerica ITC, except as provided in Section 5.1 or this Section 5.2. The provisions of Article 10 shall survive any termination of this Agreement.
- 5.2.2 Divestiture or Sale of Transferred Facilities. This Agreement shall terminate as to any Transmission Owner on the date such Transmission Owner contributes to the Company pursuant to Section 3.1(b) of the LLC Agreement or otherwise sells to a Person other than the Company any of such Transmission Owner's Transferred Facilities, but such termination shall only be to the extent of and extend to such Transmission Owner only in respect of such contributed or otherwise sold Transferred Facilities. Any proposed sale by a Transmission Owner of Transferred Facilities shall be conditioned on the delivery by the purchaser thereof of an assumption agreement in form and substance reasonably satisfactory to the Company pursuant to which the purchaser assumes the obligations of the selling Transmission Owner hereunder. Such Transmission Owner shall remain a party hereto to the extent such Transmission Owner continues to own Transferred Facilities. In the event that a Transmission Owner has contributed all of its Transferred Facilities to the Company or has otherwise sold all of its Transferred Facilities, this Agreement shall

terminate as to the Transmission Owner as of the effective date of the contribution or sale (as the case may be) of all of such Transmission Owner's Transferred Facilities, including in the case of a Transmission Owner that retains ownership of its Non-transferred Facilities, except with respect to any obligations of the Transmission Owner hereunder accruing prior to the effective date of the contribution or sale, including indemnification obligations pursuant to Section 4.2.4 arising from any event occurring prior to the effective date of the contribution or sale.

- **5.2.3 ITC Status.** A Transmission Owner may terminate this Agreement with respect to the Transferred Facilities at any time following (i) a determination by the Commission that the Company has ceased to be an ITC, or (ii) an order by the Commission requiring or authorizing the Company or the Transferred Facilities to become subject to the control or direction of a Regional Transmission Organization other than the Midwest ISO.
- **5.2.4** Failure to Achieve Transmission Service Date. If the Transmission Service Date has not occurred on or before June 30 2003, then at any time after June 30, 2003, any Party may, upon thirty (30) days prior written notice to the other Parties, cause the GridAmerica ITC and this Agreement to terminate.
- **5.3** Effect Of Termination Pursuant to Sections 5.1 or 5.2.3. In the event that a Transmission Owner terminates this Agreement pursuant to Sections 5.1 or 5.2.3, the following provisions shall apply.
 - **5.3.1** Transmission Customers Held Harmless. Transmission Customers and Interconnection Customers taking service that involves a terminating Transmission Owner's Transferred Facilities, including Transmission Service that involves transmission service agreements and interconnection and/or operating agreements executed before such Transmission Owner provided notice of its termination shall receive service for the remaining term of the agreement at the same rates, terms, and conditions that would have been applicable if there were no termination, unless such agreements are modified by the Commission in accordance with its statutory authority or by agreement of the parties to the agreement. Such Transmission Owner shall make its facilities available to provide service to such Transmission Customers and Interconnection Customers and allow the Company to continue to exercise Functional Control over the Transferred Facilities with respect to such service and shall receive no more in revenues for that service than if there had been no termination by such Transmission Owner. This Section 5.3.1 shall survive the termination of the Agreement by a Transmission Owner.
 - **5.3.2** Existing Obligations. Obligations incurred and payments applicable to time periods prior to the effective date of such termination shall survive

the termination of this Agreement. The reconciliation and payment of all such amounts shall be done as soon as practicable following such termination. This <u>Section 5.3.2</u> shall not constitute a general guarantee of the obligations of the Company by any Transmission Owner.

- 5.3.3 Construction of Facilities. Obligations relating to the construction of new facilities pursuant to an approved plan of the Company, which imposed duties upon the terminating Transmission Owner prior to such Transmission Owner's termination, shall be renegotiated as between the Company and the terminating Transmission Owner taking into consideration commitments made to Transmission Customers and Interconnection Customers and funds advanced, if any, prior to termination of this Agreement with respect to such terminating Transmission Owner. Such obligations shall survive the termination of this Agreement with respect to such Transmission Owner.
- **5.3.4** Other Obligations. Other obligations between the Company and the terminating Transmission Owner shall be renegotiated as between the Company and the terminating Transmission Owner, subject to approval by the Commission. If such obligations cannot be resolved through negotiations, they shall be resolved in accordance with the dispute resolution procedures provided for in this Agreement.
- **5.4** Effect of Termination Pursuant to Section 5.2.2. In the event that a Transmission Owner terminates this Agreement pursuant to Section 5.2.2 by divesting to the Company or otherwise selling to another Person all of its Transferred Facilities, the following provisions shall apply.
 - **5.4.1** Transmission Customers Held Harmless. In the case of a Transmission Owner that retains ownership of its Non-transferred Facilities, Transmission Customers taking service that involves transmission service over such Transmission Owner's Non-transferred Facilities pursuant to agreements executed before such Transmission Owner provided notice of its termination shall continue to receive service for the remaining term of the agreement at the same rates, terms, and conditions that would have been applicable if there were no withdrawal, unless such agreements are modified by the Commission in accordance with its statutory authority or by agreement of the parties to the agreement. Such Transmission Owner shall make its Non-transferred Facilities available to provide service to such Transmission Customers and allow the Company to continue to exercise Functional Control over the Non-transferred Facilities and shall receive no more in revenues for that service than if there had been no termination by such Transmission Owner. This Section 5.4.1 shall survive the termination of the Agreement by a Transmission Owner.
 - **5.4.2** Existing Obligations. Obligations incurred and payments applicable to time periods prior to the effective date of such termination shall survive

the termination of this Agreement. The reconciliation and payment of all such amounts shall be done as soon as practicable following such withdrawal. This <u>Section 5.4.2</u> shall not constitute a general guarantee of the obligations of the Company by any Transmission Owner.

- **Other Obligations.** Other obligations between the Company and the terminating Transmission Owner shall be renegotiated as between the Company and the terminating Transmission Owner, subject to approval by the Commission. If such obligations cannot be resolved through negotiations, they shall be resolved in accordance with the dispute resolution procedures provided for in this Agreement.
- 5.5 <u>Regulatory And Other Approvals Or Procedures.</u> The termination by a Transmission Owner of its obligations hereunder and/or withdrawal of some or all of its Transferred Facilities from the GridAmerica ITC shall be subject to applicable federal and state regulatory approvals or other regulatory procedures, including, without limitation, any required approval of the Commission and any requirement of the Commission that such Transmission Owner join a Commission approved regional transmission organization.

ARTICLE VI DISPUTE RESOLUTION

- 6.1 Negotiations. If a dispute between any two or more Parties arises out of or relates to this Agreement, any such Party may notify each other Party that it intends to initiate the dispute resolution procedures set forth herein. Immediately upon the receipt of such notice, the Party sending the notice and each other Party receiving the notice shall refer such dispute to a senior executive officer (the "SEOs") of each such Party for consultation and advice prior to the commencement of the arbitration proceedings. The SEOs shall meet in person or by teleconference as soon as mutually practicable to consider such matters. If the SEOs fail to resolve such dispute within thirty (30) days of such notice being sent, any Party to the dispute may declare the consultation procedure set forth in this Section 6.1 terminated and refer the dispute or controversy to arbitration pursuant to Section 6.2.
- 6.2 Arbitration. If a dispute between any two or more Parties arises out of or relates to this Agreement or to the relationship between the Parties created by this Agreement, and such Parties have not successfully resolved such dispute through negotiation on or before the thirtieth (30th) day following the notice referred to in Section 6.1, then such dispute shall be resolved according to this Section 6.2. If such dispute is subject to the jurisdiction of the Commission, then any Party to the dispute may, within sixty (60) days of the notice referred to in Section 6.2, bring such dispute before the Commission for resolution. If no Party brings the dispute before the Commission within sixty (60) days of the notice referred to in Section 6.1, or if the dispute is not subject to the jurisdiction of the Commission, then such dispute shall be resolved by binding arbitration ("Arbitration") under the following provisions. For the avoidance of doubt, this Section 6.1 does not apply to disputes arising under the MISO OATT, which shall be resolved in accordance with the procedures set forth therein, or to disputes arising under another agreement between and among the Company and one or more of the Transmission Owners or between the

Company and the Midwest ISO, which shall be resolved in accordance with the dispute resolution procedures of such other agreement.

- All Claims To Be Arbitrated. Except as provided in the (a) immediately preceding sentence and in Sections 6.2(1) and 9.2.1, any and all claims, counterclaims, demands, causes of action, disputes, controversies and other matters in question arising out of or relating to this Agreement, any provision hereof, the alleged breach hereof, or in any way relating to the subject matter hereof or the relationship between the Parties created hereby, involving the Parties ("Claims"), shall be finally resolved by binding arbitration by a panel of arbitrators under the Commercial Arbitration Rules (the "Arbitration Rules") of the American Arbitration Association (the "AAA") to the extent not inconsistent with the provisions of this Agreement, regardless of whether some or all of such Claims allegedly (i) are extra-contractual in nature, (ii) sound in contract, tort, or otherwise, (iii) are provided by federal or state statute, common law or otherwise or (iv) seek damages or any other relief, whether at Law, in equity or otherwise.
- (b) Referral of Claims to Arbitration. Subject to Section 6.1, one or more Parties may refer a Claim to arbitration (the "Claimant Party") by providing notice (an "Arbitration Notice") to each other Party or Parties against which the Claim is asserted (whether one or more parties, the "Respondent Party") in the manner set forth in the Arbitration Rules. The Arbitration Notice must include a general description of the Claim and shall identify all Respondent Parties and the reasons for asserting the Claim against each Respondent Party. The Arbitration is commenced between the Claimant Party and the Respondent Party ("Dispute Parties") by sending the Arbitration Notice to the Respondent Party.
- Commission Proceedings; Effect of Commission Orders. Following commencement of the Arbitration, if a Party other than a Dispute Party institutes a proceeding before the Commission that involves one or more of the Dispute Parties and the relief sought in that proceeding would require the Commission to resolve one or more issues presented in the Arbitration (a "Related Proceeding"), then the Dispute Parties agree that the Arbitration shall be stayed during the pendency of such Related Proceedings. The Dispute Parties further agree that the Commission's resolution in Related Proceedings of any issue that is also presented in the Arbitration shall be and is final and binding as to that issue in the Arbitration.

- (d) <u>Number and Qualification of Arbitrators</u>. The panel of arbitrators (the "<u>Panel</u>") shall consist of three arbitrators appointed in accordance with this <u>Section 6.2</u> and the Arbitration Rules. Arbitrators shall meet the qualifications for arbitrators established by the AAA and, in addition, shall have significant experience in the electric industry and/or significant experience as an arbitrator in complex commercial matters. The chairperson shall take an oath of neutrality.
- (e) Appointment of Arbitrators. By the fifteenth (15th) day following the day on which the Arbitration Notice is sent to the Respondent Party, the Claimant Party shall submit its appointment of the first arbitrator to the Respondent Party and the AAA. If the Claimant Party consists of more than one Party, then those Parties shall jointly appoint the first arbitrator. By the fifteenth (15th) day following the appointment of the first arbitrator, the Respondent Party shall submit its appointment of the second arbitrator to the Claimant Party and the AAA. If the Respondent Party consists of more than one Party, then those Parties shall jointly appoint the second arbitrator. The two arbitrators appointed by the Dispute Parties shall appoint a third arbitrator, who shall be the chairperson of the Panel, by the fifteenth (15th) day following the appointment of the second arbitrator. If the second arbitrator has not been appointed by the fifteenth (15th) day following the appointment of the first arbitrator, or if the first two arbitrators have not appointed the third arbitrator by the fifteenth (15th) day following the appointment of the second arbitrator, any Dispute Party may request the AAA to appoint the arbitrator(s) in question. If any arbitrator resigns, becomes incapacitated, or otherwise refuses or fails to serve or to continue to serve as an arbitrator, the Dispute Party or arbitrators entitled to designate that arbitrator shall promptly designate a successor. In the event that either of the Claimant Party or the Respondent Party consist of more than one Party and those Parties are unable to agree on the appointment of an arbitrator, then all three arbitrators shall be appointed by the AAA; provided, however, that the arbitrators so appointed shall meet the qualifications set forth in Section 6.2(d).
- **Governing Law.** In deciding the substance of the Parties' Claims, the arbitrators shall first rely upon the provisions of this Agreement and shall then apply the substantive laws governing this Agreement pursuant to Section 10.2.
- (g) <u>Powers of the Arbitrators; Limitations On Remedies</u>. The validity, construction and interpretation of this Agreement to arbitrate, and all procedural aspects of the arbitration conducted pursuant to this Agreement to arbitrate, including the

determination of the issues that are subject to arbitration (i.e., arbitrability), the scope of the arbitrable issues, allegations of "fraud in the inducement" to enter into this Agreement or this arbitration provision, allegations of waiver, laches, delay or other defenses to arbitrability, and the rules governing the conduct of the arbitration (including the time for filing an answer, the time for the filing of counterclaims, the times for amending the pleadings, the specificity of the pleadings, the extent and scope of discovery, the issuance of subpoenas, the times for the designation of experts, whether the arbitration is to be stayed pending resolution of related litigation involving third parties not bound by this arbitration agreement, the receipt of evidence and the like), shall be decided by the arbitrators to the extent not provided for in this Article VI. The arbitrators shall decide the Claims based on this Agreement, the Arbitration Rules, and the governing Law, and not ex aqueo et bono, as amiable compositeurs, or in equity. The arbitrators shall not have the power to award any of those remedies which are precluded by Section 9.2. The arbitrators shall also have the power to enter such interim orders as they deem necessary, including orders to preserve the subject matter of the Claim or to preserve or adjust the status of the Parties pending resolution of the Claim in the Arbitration. The chairperson is empowered to issue interim orders on his own authority in emergency situations and where necessary to ensure the efficient administration of the Arbitration on application from a Dispute Party, which orders shall remain in effect until a meeting of all arbitrators may be convened to consider the application. The arbitrators shall have the power to assess the attorneys' fees, costs and expenses of the Arbitration (including the arbitrators' fees and expenses) against one or more of the Parties in whatever manner or allocation the arbitrators deem appropriate.

(h) Venue; Procedural Issues. The seat of the Arbitration shall be New York, New York, or such other place as the Dispute Parties may agree. The arbitrators shall set the date, the time and the place of the hearing, which must commence on or before the one hundred twentieth (120th) day following the designation of the third arbitrator. All decisions of the three arbitrators shall be made by majority vote. In determining the extent of discovery, the number and length of depositions and all other pre-hearing matters, the arbitrators shall endeavor, to the extent possible, to streamline the proceedings and minimize the time and cost of the proceedings. There shall be no transcript of the hearing. The final hearing shall not exceed ten (10) Business Days, with the Claimant Party and Respondent Party each granted one-half of the allocated time to present its case to the arbitrators. All proceedings conducted hereunder and the decision of the arbitrators shall be kept

- confidential by the arbitrators, the AAA and any Persons participating in the Arbitration, except that the confidentiality obligations of the Parties shall be governed by <u>Section 10.14</u>.
- **(i)** Additional Claims. After the Arbitration has commenced and the Panel has been appointed, if a further Claim arises under this Agreement that is not successfully settled pursuant to Section 6.1, and the further Claim (an "Additional Claim") is related to the Claim in the Arbitration or involves the same Dispute Parties, then any Party to the Additional Claim may ask the Panel to accept jurisdiction over the Additional Claim and include it in the Arbitration by submitting an Arbitration Notice in the manner set forth in Section 6.2(b) (an "Additional Arbitration Request") and submitting a concurrent request to the Panel to accept the Additional Claim. The Parties agree that the Panel should accept jurisdiction over an Additional Claim if the resolution of the Claim before the Panel will involve some or all of the same legal and factual issues presented by the Additional Claim or if accepting jurisdiction over the Additional Claim would facilitate or help minimize the costs of resolving the disputes at issue and not unduly delay the Arbitration. The Parties agree, however, that the Panel alone shall determine whether it should accept jurisdiction over an Additional Claim and that its determination shall be final and unappealable. If the Panel refuses jurisdiction over the Additional Claim, then the Additional Arbitration Request shall constitute a separate request for arbitration, which shall proceed independently and under this Section 6.2 as if filed on the date the Panel denied the request to accept jurisdiction. So long as there is no pending Additional Arbitration Request to the Panel to accept jurisdiction, any Party to an Additional Claim may commence a separate arbitration proceeding in the manner set forth in this Section 6.2.
- **Arbitration Awards.** The arbitrators shall render their award on or before the thirtieth (30th) day following the last session of the hearing fully resolving all Claims that are the subject of the Arbitration. The award shall be in writing, shall give reasons for the decision(s) reached by the arbitrators and shall be signed and dated by the arbitrators, and a copy of the award shall be delivered to each of the Dispute Parties. A Party against which the award assesses a monetary obligation or enters an injunctive order shall pay that obligation or comply with that order on or before the thirtieth (30th) calendar day following the receipt of the award or by such other date as the award may provide. Any award of the arbitrators shall be consistent with the limitations and terms of this Agreement. The arbitrators' award may be confirmed in, and

- judgment upon the award entered by, any court having jurisdiction over the Parties.
- (k) <u>Binding Nature</u>. The decisions of the arbitrators shall be final and binding on the Parties and non-appealable to the maximum extent permitted by Law.
- **(l)** Assistance of Courts. It is the intent of the Parties that the Arbitration shall be conducted expeditiously, without initial recourse to the courts and without interlocutory appeals of the arbitrators' decisions to the courts. Notwithstanding any other provision of this Agreement, however, a Party may seek court assistance in the following circumstances: (i) if a Party refuses to honor its obligations under this Agreement to arbitrate, any other Party may obtain appropriate relief compelling arbitration in any court having jurisdiction over the refusing Party, and the order compelling arbitration shall require that the arbitration proceedings take place in New York, New York, and in the manner specified herein; (ii) a Dispute Party may apply to any state or federal court having relevant jurisdiction for orders requiring witnesses to obey subpoenas issued by the arbitrators, including requests for documents; and (iii) a Party may apply at any time before or during the Arbitration to any court having relevant jurisdiction for an order preserving the status quo ante and/or evidence in anticipation of arbitration (for avoidance of doubt, preservation of the status quo ante includes an order compelling a Party to continue to fulfill an obligation under this Agreement or to refrain from taking an action that would constitute a default under this Agreement; for further avoidance of doubt, such an application to the courts is not intended to and does not constitute waiver of the right to arbitrate Claims, nor does it refer any Claim to court for decision). The Parties agree to comply with any interim order issued by the arbitrators or by the chairperson. Any and all of the arbitrators' orders and decisions, including interim orders, may be enforced by any state or federal court having jurisdiction. Each Party agrees that arbitration pursuant to this <u>Section 6.2</u> shall be the exclusive method for resolving all Claims and that it will not commence an action or proceeding, except as provided in this Section 6.2.
- 6.3 <u>Arbitration of Certain Claims Regarding Removal of Managing Member</u>. If a Super Majority of Transmission Owners have attempted to remove the Managing Member for Cause pursuant to <u>Section 4.4.3</u> of this Agreement, and the Managing Member disputes whether Cause for removal exists (a "<u>Removal Claim</u>"), then the issue of whether Cause exists immediately shall be referred to and resolved by binding arbitration ("<u>Removal Arbitration</u>") according to this <u>Section 6.3</u>. The Removal Claim shall be finally resolved by one arbitrator appointed in accordance with this <u>Section 6.3</u> and the Arbitration Rules to the extent not inconsistent with the provisions of this Agreement. The Expedited Procedures of the Arbitration

Rules shall be used unless the arbitrator determines that they would be inappropriate. The arbitrator shall take an oath of neutrality.

- **Application to Removal Claim; Relation to Other Claims.** Any dispute other than a Removal Claim must be resolved in a separate Arbitration pursuant to Section 6.2. A Removal Arbitration may not be joined to or consolidated with an Arbitration without the consent of all parties in the Removal Arbitration and the Arbitration(s). The decision of the arbitrator on a Removal Claim shall be final and conclusive and bind any arbitrators in an Arbitration commenced under Section 6.2.
- **Referral of Claims to Arbitration.** A Managing Member who receives a 6.3.2 written notice of removal as contemplated in Section 4.4.3 (a "Removal Notice"), and who disputes that Cause for removal exists, or a Member or a Transmission Owner upon receipt of notice from the Managing Member that it disputes that Cause exists (the "Removal Claimant") may refer a Removal Claim to Removal Arbitration by providing notice (a "Notice of Removal Dispute") to the Managing Member, all Members, and all Transmission Owners that are not the Removal Claimant (whether one or more parties, the "Removal Respondent Party"), in the manner set forth in the Arbitration Rules. The Notice of Removal Dispute also must contain a list of five (5) proposed arbitrators. The Removal Arbitration is commenced between the Removal Claimant and the Removal Respondent Party ("Removal Dispute Parties") by sending the Notice of Removal Dispute to the Removal Respondent Party.
- 6.3.3 Appointment of Arbitrator. Within ten (10) days of delivery of the Notice of Removal Dispute, the Removal Respondent shall deliver to the Removal Claimant and the AAA a list of five (5) proposed arbitrators. If the lists provided by the Removal Claimant and the Removal Respondent both contain a common proposed arbitrator, such person shall be selected as arbitrator; otherwise, the AAA shall appoint one of them as arbitrator. If no persons are named on both lists, then the AAA shall appoint the arbitrator according to the procedures contained in the Arbitration Rules. If the arbitrator resigns, becomes incapacitated, or otherwise refuses or fails to serve or to continue to serve as an arbitrator, the Removal Dispute Parties shall promptly designate a successor using the procedures established in this Section 6.3. An arbitrator appointed pursuant to this Section 6.3.3 may not also be appointed as an arbitrator pursuant to Section 6.2.
- **6.3.4 Governing Law.** In deciding the substance of the Removal Claims, the arbitrator shall first rely upon the provisions of this Agreement and shall then apply the substantive laws of the State of New York, in accordance with Section 10.2 hereof.

- 6.3.5 Powers of the Arbitrators; Limitations On Remedies. The arbitrator in a Removal Arbitration shall decide solely the Removal Claim, and shall have no power to decide any other Claim. The arbitrator shall decide the Removal Claim based on this Agreement, the Arbitration Rules, and the governing law, and not *ex aqueo et bono*, as *amiable compositeur*, or in equity. The arbitrator shall have the power to assess attorney's fees (in accordance with Section 10.19), costs, and expenses of the Removal Arbitration (including the attorneys' fees and expenses) against one or more of the Parties in whatever manner or allocation the arbitrator deems appropriate.
- 6.3.6 <u>Venue</u>; <u>Procedural Issues</u>. The seat of the Removal Arbitration shall be New York, New York, or such other place as the Removal Dispute Parties may agree. The arbitrator shall set the date, the time and the place of the hearing, which must commence on or before the thirtieth (30th) day following the appointment of the arbitrator. There shall be no transcript of the hearing. The final hearing shall not exceed ten (10) Business Days, with the Removal Claimant and Removal Respondent Party each granted one-half of the allocated time to present its case to the arbitrator. All proceedings conducted hereunder and the decision of the arbitrator shall be kept confidential by the arbitrator, the AAA and any Persons participating in the Removal Arbitration.
- 6.3.7 Arbitration Awards. The arbitrator shall render his award on or before the tenth (10th) day following the hearing(s) on the Removal Claim. The award shall be in writing and shall give a reasonably detailed description of the reasons for the decision(s) reached by the arbitrator and shall be signed and dated by the arbitrator, and a copy of the award shall be delivered to each of the Removal Dispute Parties. Any award of the arbitrator shall be consistent with the limitations and terms of this Agreement. The arbitrator's award may be confirmed in, and judgment upon the award entered by, any court having jurisdiction over the Parties.

ARTICLE VII TAX MATTERS

- **7.1** Responsibility for Transmission Owner Taxes. Each Transmission Owner shall be responsible for preparation and filing of all Tax Returns and other filings related to its Transmission Business and Transferred Facilities and for payment of any Tax liabilities related to its Transmission Business and Transferred Facilities. No Party shall be responsible for, or required to, file any Tax Returns or other reports for any other Transmission Owner and shall have no liability for any Taxes related to any other Transmission Owner's Transmission Business or Transferred Facilities.
- **7.2** Responsibility for the Company Taxes. The Company shall be responsible for preparation and filing of all Tax Returns and other filings related to its business and the facilities owned by it constituting part of the Transmission System and for payment of any Tax liabilities

related thereto. No Transmission Owner shall have any responsibility for filing, or be required to file, any Tax Returns or other reports for the Company and shall have no liability for any Taxes related to the Company's owned facilities.

ARTICLE VIII FORCE MAJEURE

If a Party is rendered wholly or partly unable to perform its obligations under this Agreement because of a Force Majeure event, that Party shall be excused from whatever performance is affected by such Force Majeure event but only to the extent so affected, provided that: (i) the Party, within two (2) days after the occurrence of the Force Majeure event, gives the other Parties notice describing the particulars of the occurrence and its estimated duration; (ii) the suspension of performance is of no greater scope and of no longer duration than is required by the Force Majeure event; and (iii) the Party uses commercially reasonable efforts to remedy its inability to perform and to resume its full performance under this Agreement; provided, however, that the Party's obligation to remedy its inability to perform shall not require the settlement of any strike, walkout, lockout or other labor dispute on terms that, in the sole judgment of the Party involved in said dispute, are contrary to its best interest.

ARTICLE IX BREACH, CURE AND DEFAULT

9.1 <u>Continued Operation.</u> In the event of a breach or default hereunder by any Party, the Parties shall continue to operate and maintain their respective facilities and perform all acts reasonably necessary for the Company to operate and maintain the Transmission System in accordance with this Agreement and the MISO ITC Agreement.

9.2 Remedies.

- Party of this Agreement will result in irreparable damage to the other Parties for which no money damages could adequately compensate. In addition to all other remedies to which the other Parties may be entitled hereunder, including reasonable attorneys' fees pursuant to Section 10.19 and court costs, any other Party shall be entitled to seek injunctive relief or specific performance to restrain or compel the breaching Party. Each of the Parties expressly waives any claim that an adequate remedy at Law exists for such a breach.
- **9.2.2** All Other Remedies. No right or remedy herein conferred is intended to be exclusive of any other available right or remedy, but each and every such right or remedy shall be cumulative and shall be in addition to every other right or remedy given hereunder or hereafter existing under Law or in equity. The exercise of any one right or remedy shall not be deemed an election of such right or remedy or preclude the exercise of any other right or remedy. The resort to any right or remedy provided for herein or provided for by Law or in equity shall not prevent the concurrent or

subsequent employment of any other right or remedy. Notwithstanding the foregoing:

- (a) as between the Parties, notwithstanding anything to the contrary in this Agreement, no Party will be liable to any other Party for indirect, consequential, special or punitive damages on account of any action or proceeding brought hereunder or related hereto; provided, however, that the foregoing shall not apply to indemnity obligations of any Party hereunder that relate to liabilities of the Indemnified Party to third parties, even if such liabilities to third parties include liability for indirect, consequential, special or punitive damages suffered by such third parties; and
- (b) except with respect of the obligations of the Company (i) to reimburse Transmission Owners for costs on a Pass-Through Basis under the circumstances described in Sections 2.3, 2.4, 2.5, 4.1.3(a), (ii) to pay amounts to the Transmission Owners on a Pass-Through Basis as provided in this Agreement, (iii) to perform its obligations under Section 3.3.4, and (iv) to indemnify the Transmission Owners as provided in Section 3.4.1, the Company shall be liable to the Transmission Owners only in the case of Gross Negligence or Willful Misconduct.
- 9.3 <u>No Joint Liability</u>. The liabilities and obligations of each Party under this Agreement shall be several and not joint, and no Party shall be jointly liable under this Agreement with any other Party. Without in any way limiting the foregoing, the failure of one Transmission Owner to perform its obligations hereunder shall not constitute a failure to perform by any other Transmission Owner.

ARTICLE X MISCELLANEOUS PROVISIONS

- **10.1** Not for Benefit of Third Parties. This Agreement is intended to be solely for the benefit of the Parties, their successors and permitted assignees, and is not intended to and shall not confer any rights or benefits on any Person not a signatory hereto.
- 10.2 Governing Law. THE VALIDITY AND INTERPRETATION OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF NEW YORK, EXCEPT TO THE EXTENT THAT THIS AGREEMENT IS REQUIRED TO BE INTERPRETED OR ENFORCED IN ACCORDANCE WITH THE FEDERAL LAW OF THE UNITED STATES OF AMERICA, AND WITHOUT REFERENCE TO THE CHOICE OF LAW PRINCIPLES IN THE STATE OF NEW YORK THAT REQUIRE APPLICATION OF THE LAWS OF A DIFFERENT JURISDICTION.
- 10.3 <u>Successors And Assigns</u>. This Agreement shall inure to the benefit of, and be binding upon, each of the Parties, their respective successors and assigns. This Agreement may not be assigned by a Transmission Owner except that this Agreement (i) shall be assigned to any

successor to the ownership of a Transmission Owner's Transferred Facilities, and any such successor shall become a Transmission Owner under this Agreement or (ii) may be assigned to a lender. This Agreement may not be assigned, by operation of Law or otherwise by the Company, except to a lender. An assignment of this Agreement by a Transmission Owner as to its Transferred Facilities in conformance with clause (i) of this Section 10.3 shall release such Transmission Owner from further liability or obligation under this Agreement as to obligations arising after the date of such assignment.

- **10.4 Effect of Waiver.** No waiver by a Party of any one or more defaults by another Party hereto in the performance of this Agreement shall operate or be construed as a waiver of any future default or defaults, whether of a like or different character.
- 10.5 <u>Severability</u>. Except for <u>Section 5.1</u> of this Agreement, any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of that prohibition or unenforceability without invalidating the remaining provisions hereof or affecting the validity or enforceability of that provision in any other jurisdiction, and such invalid, void, or unenforceable provision may be replaced with a suitable and equitable provision pursuant to <u>Section 10.6</u> in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid, void, or unenforceable provision.
- **10.6** Renegotiation. If any provision of this Agreement, or the application thereof to any Person, Entity, or circumstance, is held by a court or Governmental Authority of competent jurisdiction to be invalid, void, or unenforceable, or if a modification or condition to this Agreement is imposed by a Governmental Authority exercising jurisdiction over this Agreement, then the Parties shall endeavor in good faith to negotiate such amendment or amendments to this Agreement as will restore the relative rights, obligations, and economic position of the Parties under this Agreement immediately prior to such holding, modification, or condition to the extent consistent with the orders, decisions or requirements of the applicable Governmental Authority. The rights and obligations of the Parties under this Agreement shall be modified to the extent necessary (i) to comply with an order of the Commission or (ii) as is necessary to avoid a determination by the Commission that the Company has ceased to be an ITC; provided, however, that any such modification (including any modification of the definition of Functional Control and the scope of activities performed by each Party hereunder) shall preserve to the maximum extent possible the relative rights, obligations, and economic positions of the Parties hereunder as in effect prior to giving effect to such modification.
- **10.7** <u>Representations And Warranties.</u> Each Party represents and warrants, as of the Effective Date, to each other Party that:
 - (i) it is duly organized, validly existing and in good standing under the applicable Laws of the jurisdiction of its organization and is qualified to do business (A) in the jurisdictions necessary to perform this Agreement, in the case of the Company, or (B) in the jurisdictions in which its Transferred Facilities are located, in the case of a Transmission Owner;

- (ii) it has the power to execute and deliver this Agreement and to perform its obligations under this Agreement and has taken all necessary corporate and/or other actions to authorize such execution, delivery and performance;
- (iii) its execution and delivery of this Agreement and its performance of its obligations under this Agreement do not violate or conflict with (A) any Laws applicable to it, (B) any provision of its charter or by-laws (or comparable constituent documents), (C) any order or judgment of any court or Governmental Authority applicable to it or any of its assets or (D) any contractual restriction binding on or affecting it or any of its assets, except third-party joint agreements covered by Section 10.13;
- (iv) subject to the receipt of any Required Consents (as defined in the Master Agreement) required by it, to its knowledge, there are no Consents required for it and its Affiliates to perform their respective obligations under this Agreement other than Consents which have been obtained and Consents which may be required to perform obligations which, by the terms of this Agreement, will not arise and are not required to be performed except upon the happening of one or more contingencies specified in this Agreement;
- (v) this Agreement has been duly executed and delivered by the Party and constitutes its legal, valid, and binding obligation, enforceable against it in accordance with its terms, subject to applicable bankruptcy, reorganization, insolvency, moratorium or similar Laws affecting creditors' rights generally and subject, as to enforceability, to equitable principles of general application (regardless of whether enforcement is sought in a proceeding in equity or at Law); and
- (vi) except as otherwise permitted herein, it has neither initiated nor received written notice of any pending action, proceeding, or investigation, nor to its knowledge is any such action, proceeding, or investigation threatened (or any basis therefor known to it), which questions the validity of this Agreement, or which would materially or adversely affect its rights or obligations under this Agreement.
- **10.8 Further Assurances.** Each Party agrees that it shall hereafter execute and deliver such further instruments, provide all information, and take or forbear such further acts and things as may be reasonably required and useful to carry out the intent and purpose of this Agreement and as are not inconsistent with the provisions of this Agreement.

- Party pursuant to this Agreement shall be directed to such Party's representative at the address or facsimile number for such Party set forth on Appendix C or to such other address or facsimile number as the Party may designate by written notice to each other Party from time to time. All notices or other communications required or permitted to be given pursuant to this Agreement must be in writing and will be considered as properly given if sent by facsimile transmission (with confirmation notice sent by first class mail, postage prepaid), by reputable nationwide overnight delivery service that guarantees next Business Day delivery, by personal delivery, or, if mailed from within the United States, by first class United States mail, postage prepaid, registered or certified with return receipt requested. Any notice hereunder will be deemed to have been duly given (i) on the date personally delivered, (ii) when received, if sent by certified or registered mail, postage prepaid, return receipt requested or if sent by overnight delivery service; and (iii) if sent by facsimile transmission, on the date sent, provided confirmation notice is sent by first-class mail, postage prepaid promptly thereafter.
- 10.10 <u>Regulatory Proceedings</u>. Subject to the provisions of <u>Section 10.6</u>, no Party shall seek to amend this Agreement through a unilateral application to the Commission to do so. Notwithstanding the foregoing, nothing in this Agreement shall be deemed to prohibit or limit any Party's right to otherwise initiate or intervene in any proceedings before the Commission or any other Governmental Authority involving its rights, duties, and obligations under this Agreement or any aspect of the Transmission System.
- 10.11 Entire Agreement; Amendments. This Agreement constitutes the entire agreement among the Parties pertaining to the subject matter hereof and supersedes all prior agreements, representations and understandings, written or oral, pertaining thereto including that certain Letter of Intent dated June 20, 2002 among NGUSA, Ameren, ATSI and NIPSCO regarding the formation and operation of GridAmerica as an ITC within the Midwest ISO. Except as otherwise provided in Section 3.4.3, no amendment to or modification, termination or waiver of or under any provision of this Agreement shall be valid unless the same shall be in writing and signed by the Company and one or more Transmission Owners owning Transmission Facilities with a Net Plant of at least 66.67% of the aggregate Net Plant of all Transmission Facilities subject to the Functional Control of the Company pursuant hereto (other than any Transmission Facilities owned by the Company); provided, however, that (i) Sections 3.4.1, 4.2.4 and 4.3 may only be so amended, modified, terminated or waived with the unanimous written consent of the Company and all the Transmission Owners, (ii) any amendment, modification, termination or waiver that materially adversely affects a specific Transmission Owner must be approved in writing by such Transmission Owner, and (iii) a waiver by a Party as to only its rights may be granted by such Party.
- 10.12 Efforts of the Parties. Where the consent, agreement, or approval of a Party must be obtained hereunder, such consent, agreement, or approval shall not be unreasonably withheld, conditioned, or delayed. Where a Party is required or permitted to act, or omit to act, based on its opinion or judgment, such opinion or judgment shall not be unreasonably exercised. To the extent that the jurisdiction of any Governmental Authority applies to any part of this Agreement and/or the transactions or actions covered by this Agreement, each Transmission Owner shall cooperate with the Company and all other Transmission Owners to secure any

necessary or desirable approval or acceptance of such Governmental Authorities of such part of this Agreement and/or such transactions or actions.

- 10.13 Third-Party Joint Agreements. This Agreement, including the appendices, shall not be construed, interpreted, or applied in such a manner as to cause a Transmission Owner to be in material breach, anticipatory or otherwise, of any agreement in effect on the Effective Date, between such Transmission Owner and one or more third parties for the joint ownership, operation, or maintenance of any electrical facilities covered by this Agreement or the MISO OATT. If a Transmission Owner has such a third-party joint agreement, it shall discuss with the Company any material conflict between such third-party joint agreement and this Agreement raised by a third party to such joint agreement, but the resolution of such a conflict shall, vis-à-vis the Company, be and remain within the sole discretion of the Transmission Owner; provided, however, that the Transmission Owner shall, if the conflict is otherwise unresolved, utilize the available remedies and dispute resolution procedures to resolve such conflict, including, but not limited to, submitting such conflict to the Commission for resolution; provided, further, that unless so ordered by the Commission or a court having jurisdiction, in no event shall the Transmission Owner enter into a resolution of such conflict which would impair the reliability of the Transmission System. Each Transmission Owner hereby represents and warrants that, except as otherwise set forth on Appendix B, the application of the provisions of this Section 10.13 will not have a material adverse effect on the ability of the Company to perform its obligations under this Agreement. Notwithstanding anything to the contrary in this Agreement, the Company shall not be deemed to be in breach of any of its obligations under this Agreement or subject to liability hereunder if it takes action or fails to take action with respect to any matter involving a Transmission Owner's exercise of its rights under this Section 10.13.
- **10.14** Confidentiality. The following provisions set forth the obligations arising out of the disclosure of Confidential Information by a Party (the "Disclosing Party") to one or more Parties (the "Recipient" or "Recipients") under any Transaction Agreement.
 - (a) Agreement of Non Disclosure and Non-Use. In consideration of the disclosure by a Disclosing Party to a Recipient of Confidential Information, the Recipient and its officers, directors, partners, employees, Affiliates, agents, representatives, outside auditors, attorneys, and any Third Party Recipient who have access to the Confidential Information (collectively, "Representatives"):
 - shall keep Confidential Information confidential and will not, without the prior written consent of such Disclosing Party or as allowed by this Agreement, disclose Confidential Information to other Persons; and
 - (ii) shall not use Confidential Information other than for purposes legitimately related to the operation of the business of the Company, including, without limitation, the exercise of Functional Control over any Transferred Facilities ("Approved Uses"); provided, however, that

nothing contained herein shall limit the right of the Company or any of its Representatives from using any Confidential Information disclosed to the Company by any Party consisting of methods, techniques, rate design and other similar Confidential Information which relates to the electric transmission industry generally, and not to the business of any Party, (but excluding any software developed by the Company or any Transmission Owner and excluding any Confidential Information of a Transmission Owner marked "Proprietary" by such Transmission Owner) for other than Approved Uses which are not in competition with the business of such Party. Each Recipient agrees to transmit the Confidential Information of a Disclosing Party only to such of the Recipient's Representatives who need to know the Confidential Information for the purpose of assisting the Recipient in Approved Uses, and who are informed of the provisions of this Section 10.14. A Recipient shall be fully liable for any breach of this Agreement by its Representatives and agrees, at its sole expense, to take reasonable measures to restrain its Representatives from prohibited or unauthorized disclosure or use of the Confidential Information.

- **(b)** Disclosure Required by Subpoena, Law, Litigation or Legal **Process.** If any portion of Confidential Information is required to be disclosed by subpoena, Law, litigation, arbitration, or similar legal process, or to a Governmental Authority, the Recipient will promptly inform the Disclosing Party of the existence, terms and circumstances surrounding such request before any such disclosure is required so as to allow the Disclosing Party to protect the Confidential Information. The Recipient will consult with the Disclosing Party on the advisability of taking legally-available steps to resist or narrow such request. The Disclosing Party may thereafter seek to obtain a protective order, and the Recipient shall cooperate with the Disclosing Party in its efforts to obtain a protective order, to restrict access to, and any use or disclosure of, the Confidential Information, at the expense of the Disclosing Party. Notwithstanding anything else to the contrary contained herein, Confidential Information that is required to be disclosed in the ordinary course of the Company's business to the Commission or other Governmental Authority pursuant to Law or the MISO OATT may be so disclosed without compliance with this Section 10.14(b).
- (c) <u>Disclosure In Connection with Financing Transactions or</u>
 Transfer of Transferred Facilities. In the event that a Recipient

desires to disclose Confidential Information in connection with a financing or other similar transaction or in connection with a transfer of its Transferred Facilities, including as part of the due diligence requested by a proposed counterparty or transferee (a "Third Party Recipient"), such Recipient may disclose such Confidential Information to such Third Party Recipient only after receipt by such Recipient from such Third Party Recipient of a confidentiality agreement containing substantially the terms and conditions set forth in this Section 10.14; provided, however, that no competitively sensitive Confidential Information concerning a Transmission Owner may be disclosed to any Person that is a direct competitor of such Transmission Owner without such Transmission Owner's prior written consent.

- **(d)** Disclosure in Connection with Dispute. A Recipient may disclose Confidential Information to (i) the Panel in connection with an Arbitration pursuant to Section 6.2, (ii) to the Commission in connection with a Claim being heard by the Commission and (iii) to a court in connection with a dispute being heard by such court; provided, however, that the Recipient shall take reasonable steps to protect the confidentiality of such Confidential Information and, where the Recipient would not be materially adversely affected by its disclosure to the Disclosing Party of its intent to disclose such Confidential Information in connection with a dispute as provided above, the Recipient shall so disclose to the Disclosing Party the Recipient's intent to so disclose such Confidential Information so as to allow the Disclosing Party the opportunity to protect such Confidential Information. In such case, the Recipient shall cooperate with the Disclosing Party in its efforts to obtain a protective order, to restrict access to, and any other use or disclosure of, the Confidential Information.
- (e) <u>Survival of Obligations</u>. The obligations with respect to Confidential Information set forth herein shall survive the termination of this Agreement for five (5) years. Upon the termination of the obligations of this Agreement with respect to an item of Confidential Information, the Recipient shall be free to use and disclose such item of information freely and without any obligation to the Disclosing Party.
- (f) Ownership of Confidential Information. Each Disclosing Party reserves its (and, if applicable, its licensor's) ownership rights in and to its Confidential Information disclosed to a Recipient and only grants a license to use such Confidential Information for the Approved Uses. In addition, each Recipient agrees that it does not acquire any ownership interest in the Confidential Information of any Disclosing Party by virtue of the combination of such

Confidential Information with other Confidential Information, including that of the Company.

- **10.15** No Partnership. This Agreement is not intended, and shall not be construed, interpreted or applied, to create a partnership or joint venture, and no Transmission Owner shall be entitled to act as an agent for any other Transmission Owner with respect to the Company.
- **10.16** <u>Current Documents</u>. The Company shall maintain current versions of all agreements and procedures, all amendments thereto and shall post such documents on its Internet World Wide Web Site or equivalent form of electronic posting and provide such documents to the Transmission Owners.
- 10.17 <u>Late Payments.</u> If a Party does not pay within ten (10) days of the date required hereunder, all or any portion of an amount such Party is required to pay as provided in this Agreement then (i) the amount such owing Party is required to pay shall bear interest at (A) the sum of (I) a varying rate per annum that is equal to the interest rate publicly quoted by *The Wall Street Journal*, from time to time as the prime commercial or similar reference interest rate with adjustments in that varying rate to be made on the same date as any change in that rate plus (II) 2% per annum or (B) such lower rate required under applicable Law, compounded annually and (ii) a Party to which payment is due may take any action, at the cost and expense of the owing Party to obtain payment by such owning Party of the portion of such owning Party's payment that is in default, together with interest thereon as provided above.
- **10.18** <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument, notwithstanding that all of the Parties are not signatories to the original or to the same counterpart.
- **10.19** <u>Attorneys' Fees.</u> In any dispute arising hereunder, the party prevailing at final judgment shall be entitled to recover from the other party all of its reasonable attorneys' fees and costs incurred in such a proceeding, in addition to any affirmative or injunctive relief that it may receive.
- 10.20 <u>Time is of the Essence</u>. Time is of the essence of each provision of this Agreement.
- **10.21** Representatives. No later than fourteen (14) days after the execution of this Agreement by a Party, such Party shall appoint one (1) representative to serve as a point of contact for matters arising under this Agreement. No representative has the right to amend this Agreement or waive any rights under this Agreement. A Party may replace its representative at any time at its own discretion upon providing notice to the other Parties.

IN WITNESS WHEREOF, the Parties have caused their duly authorized representatives to execute this Agreement, on their respective behalves.

GridAmerica LLC

By \s\Nicholas P. Winser

Name (typed or printed): Nicholas P. Winser Title Chief Executive Officer

Union Electric Company d/b/a/ AmerenUE Central Illinois Public Service Company d/b/a/ AmerenCIPS

By: Ameren Services Company (their agent)

By \s\David A. Whiteley

Name (typed or printed): David A. Whiteley
Title Senior Vice President

American Transmission Systems, Incorporated

By \s\Stanley F. Szwed

Name $_{(typed\ or\ printed)}$: Stanley F. Szwed

Title Vice President

Northern Indiana Public Service Company

By \s\Jerry L. Godwin

Name (typed or printed): Jerry L. Godwin Title Chief Operating Officer

APPENDIX A

DESCRIPTION OF TRANSFERRED FACILITIES

345 kV Lines

O40 KV EIIICS	
LINE NAME	
Albion, South - Gibson (PSI)	Note 1
Casey, West - Breed (IMP Co.)	Note 1
Casey, West - Kansas, West	
Coffeen - Coffeen, North	
Coffeen, North - Holland, NW	
Coffeen - Pana North	
Holland - Neoga, South	
Pawnee, West - Kincaid (CE)	
Neoga, South - Casey, West	
Newton - Casey, West	
Newton - Xenia	
West Frankfort, East - Shawnee (TVA)	Note 2
West Frankfort, East - Norris City, North	
Xenia - Mt. Vernon (IP)	Note 1

230 kV Lines

LINE N	∧ME
LINEIN	AIVIE
W. E. J. ()	•
W. Frankfort - Joppa	

161 kV Lines

LINE NAME	
Joppa (EEInc) - Marion South	
N. Marblehead - Palmyra (UE)	Note 1

Note 1: This line is owned by CIPS and the company whose abbreviation is in parentheses.

Note 2: CIPS, IP and TVA own portions of this 345 kV line.

138 kV Lines

LINE NAME
Norris City - Albion, South
West Frankfort-Benton, Northwest
West Frankfort-Carbondale, Northwest
Benton, Northwest - Mt. Vernon 42nd St.
West Frankfort - West Frankfort, East
West Frankfort, East - Herrin, East
Herrin, East - Crab Orchard
Crab Orchard - Marion, South
Joppa (CIPS) - Joppa North Bus (EEInc)
West Frankfort - Muddy Muddy, Crab Orobord
Muddy - Crab Orchard Grand Tower - Carbondale, Northwest
Grand Tower - Carbondale, North
Makanda, North - Marion, S
Grand Tower - Steelville (IP)
Macomb, West - Macomb, Northeast
Macomb, Northeast - Niota
Macomb, West - Ipava, South
Meredosia, East - South Clayton
Quincy, East - South Clayton
Macomb, West - South Clayton
Murdock - Tuscola, West
Tuscola, West - Mattoon, West
Murdock - Kansas, West
Kansas, West - Mattoon, East
Mattoon, East - Mattoon, West
Mattoon, West - Neoga, South
Kansas, West - Hutsonville
Taylorville, South - Pana, North
Pana, North - Ramsey, East
Pana, North - Shelbyville, south Ramsey, Ease - Kinmundy
Kamsey, Ease - Killmundy Kinmundy - Louisville, South
Effingham, 138 kV - Newton
Effingham, 138 kV - Neoga, South
Shelbyville - South - Neoga, South
Neoga, South - Hutsonville
Newton - Louisville
Louisville - Olney, N
Olney, N - Albion, South
Newton - Marathon Oil Co.
Lawrenceville, South - Marathon Oil Co.
Hutsonville - Marathon Oil Co.
Olney, North - Lawrenceville, South
Lawrenceville, south - Vincennes (PSI)

Ipava, South - Havana (IP)
Ipava, South - Meredosia, East
Ipava, South - Canton, South
Marblehead, North - Quincy, South
Quincy, East - Quincy, South
Meredosia, East - Jacksonville Ind. Park
Auburn, North - Jacksonville Ind. Park
Auburn, North - Pawnee, West
Pawnee, West - Jerseyville, Northwest
Meredosia, East - Jerseyville, Northwest
Pawnee, West - Taylorville, south
Watseka - Paxton, East
Watseka - Goodland (NIPSCo)
Paxton, East - Sidney (IP)
Sidney (IP) - Rantoul Junction
Paxton, East - Gibson City
Gibson City - Brokaw (IP)
Rantoul Junction - Rantoul
Rantoul - Rising (IP)
Rantoul Junction - Murdock
Paxton, East - Hoopeston, West
Hoopeston - Vermilon (IP)
Meredosia, East - Quincy, East

138 kV Lines

LINE NAME Auburn, North - Jacksonville Ind. Park Auburn, North - Pawnee, West Benton, Northwest - Mt. Vernon 42nd St. Crab Orchard - Marion, South Effingham, 138 kV - Neoga, South Effingham, 138 kV - Newton Gibson City - Brokaw (IP) Grand Tower - Carbondale, Northwest Grand Tower - Makanda, North Grand Tower - Steelville (IP) Herrin, East - Crab Orchard Hoopeston - Vermilon (IP) Hutsonville - Marathon Oil Co. Ipava, South - Canton, South Ipava, South - Havana (IP) Ipava, South - Meredosia, East Joppa (CIPS) - Joppa North Bus (EEInc) Kansas, West - Hutsonville Kansas, West - Mattoon, East Kinmundy - Louisville, South Lawrenceville, South - Marathon Oil Co. Lawrenceville, south - Vincennes (PSI) Louisville - Olney, N Macomb, Northeast - Niota Macomb, West - Macomb, Northeast Macomb, West - South Clayton Macomb, West - Ipava, South Makanda, North - Marion, S Marblehead, North - Quincy, South Mattoon, East - Mattoon, West Mattoon, West - Neoga, South Meredosia, East - Jacksonville Ind. Park Meredosia, East - Jerseyville, Northwest Meredosia, East - Quincy, East Meredosia, East - South Clayton Muddy - Crab Orchard Murdock - Kansas, West Murdock - Tuscola, West

138 kV Lines

LINE NAME	
leoga, South - Hutsonville	
lewton - Louisville	
lewton - Marathon Oil Co.	
lorris City - Albion, South	
lney, N - Albion, South	
Olney, North - Lawrenceville, South	
ana, North - Ramsey, East	
ana, North - Shelbyville, south	
awnee, West - Jerseyville, Northwest	
awnee, West - Taylorville, south	
axton, East - Gibson City	
axton, East - Hoopeston, West	
axton, East - Sidney (IP)	
Quincy, East - Quincy, South	
uincy, East - South Clayton	
amsey, Ease - Kinmundy	
tantoul - Rising (IP)	
antoul Junction - Murdock	
antoul Junction - Rantoul	
helbyville - South - Neoga, South	
idney (IP) - Rantoul Junction	
aylorville, South - Pana, North	_
uscola, West - Mattoon, West	
Vatseka - Goodland (NIPSCo)	
Vatseka - Paxton, East	
Vest Frankfort - Muddy	
Vest Frankfort - West Frankfort, East	
Vest Frankfort, East - Herrin, East	
Vest Frankfort-Benton, Northwest	
Vest Frankfort-Carbondale, Northwest	

345 kV Lines

LINE NAME	
Bland-Franks-3	
Cahokia-Roxford-4	
Callaway-Bland-1	
Joppa 345-Kelso-1	
Kelso-Lutesville-1	
Labadie-Bland-3	
Labadie-Montgomery-4	
Labadie-Tyson-1	
Labadie-Tyson-2	
Lutesville-St. Francois-1	
Mason-Labadie-3	
Mason-Labadie-4	
Mason-Sioux-7	
Montgomery-Callaway-7	
Montgomery-Callaway-8	
Montgomery-Spencer Creek-1	
Montgomery-Overton-5 (Note1)	
Palmyra Tap-Palmyra(AECI)	
Palmyra Tap-Substation T-1	
Rush Island-St. Francois-1	
Rush Island-Tyson-1	
Rush Island-Tyson-2	
Spencer Creek-Palmyra Tap	
Shawnee(TVA)-Joppa 345-1	Note 2
Sioux-Montgomery-6	
Substation T-Hills-6	
Sioux-Roxford-5	

230 kV Lines

LINE NAME
Cahokia - Pinkneyville-1
Pinkneyville - West Frankfort-1

161 kV Lines

LINE NAME	
Adair-Appanoose(MEC)-2	Note 2
Barnett(AECI)-California-1	
Big Creek(AECI) - Warrenton -3	Note 2
California-Overton-1	
Cape-Joppa(EEI)-1	
Cape-Kelso-2	
Cape - Kelso - 3	
Clark - Cominco - 2	
Clark - Pilot Knob - 1	
Eldon - Barnett(AECI) - 1	
Fletcher (AECI)- Cominco-1	
Fletcher (AECI)- Pilot Knob-2	
Fletcher (AECI)- Sweetwater-3	
General Motors - Point Prairie-3	
Guthrie - Overton - 3	
Kelso - Miner - 2	
Kelso - Stoddard - 3	
,	Note 2
Miner - Sikeston (SPA)- 1	
Missouri City(KCLP)-Maurer Lake-5	Note 2
Montgomery - Guthrie - 3	
Osage - Eldon - 1	
Overton - Moberly - 1	
	Note 2
	Note 2
Palmyra - Viele (ALTW)- 2	
Pike - Palmyra - 2	
Point Prairie-Big Creek(AECI)-3	
Salisbury (KCP&L) -Moberly-10	Note 2
Thomas Hill(AECI)-Adair-1	
Troy(AECI) - Pike - 1	
Warrenton - Montgomery - 3	

161/138 kV Lines

LINE NAME	
Belleau - General Motors - 3	Dardenne 138/161 kV transformer in series with line
Belleau - Troy(AECI) - 1	Dardenne 138/161 kV transformer in series with line
Rivermines-Cape-1	Fredericktown 138/161 kV transformer in series with line

Note 1: McCredie Tap on Montgrmery-Overton-5 is a tie with AECI

Note 2: This line is owned by UE and the company whose abbreviation is in parentheses.

138 kV Lines

138 kV Lines

LINE NAME		LINE NAME
Arnold-Meramec-3		Mason-Meramec-1
Ashley-Venice-2		Mason-Meramec-2
Belleau - General Motors - 3		McClay-Belleau-1
Belleau - Pike - 1		Meramec-Baumgartner-1
Baumgartner-Watson-1		Meramec-Sandy Creek-1
Berkeley-Sioux-1		Meramec-St. Francois-2
Berkeley-Sioux-2		Meramec-Watson-2
Buck Knob-Selma-2		Page-Berkeley-1
Cahokia-Ashley-2		Page-Berkeley-2
Cahokia-Central-1		Page-Euclid-1
Cahokia-Central-2		Page-Mason-1
Cahokia-Dupo Ferry(IP)-1	Note 2	Page-Mason-2
Cahokia-Meramec-1	<u> </u>	Page-Sioux-4
Cahokia-Meramec-2		Page-Warson-5
Cahokia-Ridge-1	<u> </u>	Page-Warson-6
Cahokia-Ridge-2		Rivermines-Cape-1
Campbell-Euclid-4		Rivermines-Clark-2
Campbell-Maline-1		Rivermines-Maries-1
Campbell-Maline-2		St. Francois - New Bourbon-1
Central-Watson-1		St. Francois-Rivermines-1
Clark-Osage-2		St. Francois-Rivermines-2
Clark-Pea Ridge-3		St. Genevieve-Rivermines-1
Conway-Tyson-3		Sandy Creek-St. Francois-1
Conway-Tyson-4		Selma-Rivermines-2
Dupo Ferry(IP)-Buck Knob-1	Note 2	Sioux-Huster-1
East Quincy-Hamilton-4		Sioux-Huster-3
Franklin-Tegeler-1		Sioux-Maline-3
Grand Tower-Perryville-1	<u> </u>	Sioux-Maline-4
Gray Summit-Franklin-1		Sioux-Roxford-1
Gray Summit-Franklin-2		Sioux-Roxford-2
Hall Street-Ashley-1		Sioux-South Quincy-4
Huster-McClay-1		South Quincy-East Quincy-4
Huster-Belleau-3		Tegeler-Osage-1
Maries-Osage-1		Tyson-Arnold-3
Marshall-Lakeshire-4		Tyson-Meramec-4
Marshall-Tyson-1		Venice-Campbell-1
Marshall-Tyson-2	_	Venice-Campbell-2
Mason-Gray Summit-1		Venice-Ridge-4
Mason-Gray Summit-2		Warson-Mason-3
Mason-McClay-9		Warson-Mason-4

Note 2: This line is owned by UE and the company whose abbreviation is in parentheses.

Substations at Production Plants

SUBSTATION NAME	For Specified Voltage(s) Only Ky
CALLAWAY PLANT	345
LABADIE PLANT	345
MERAMEC PLANT	138
OSAGE PLANT	138
RUSH ISLAND PLANT	345
SIOUX PLANT	345/138
VENICE PLANT	138

Substations - Voltage Tie

	For Specified Voltages Only
SUBSTATION NAME	KV
BELLEAU	345/138
CAHOKIA	138/230/345
CAMPBELL	345/138
CLARK	161/138
DARDENNE	138/161
FREDERICKTOWN	161/138
GRAY SUMMIT	345/138
KELSO	345/161
MASON	345/138
MONTGOMERY	345/161
OSAGE	138/161
OVERTON	345/161
ROXFORD	345/138
ST.FRANCOIS	345/138
TYSON	345/138

Substations - Switching and Bulk

	For Specified
	Voltage Only
SUBSTATION NAME	kV
ADAIR	161
ARNOLD	138
ASHLEY	138
BAUMGARTNER	138
BERKELEY	138
BLAND	345
BLANCHETT	138
BRUSHY CREEK	161
BUCK KNOB	138
CALIFORNIA	161
CAMPBELL	345
CAPE GIRARDEAU	161
CENTRAL	138
COMINCO MINE	161
CONWAY	138
EAST QUINCY	138
ELDON	161
EUCLID	138
FRANKLIN	138
GASCO	138
GENERAL MOTORS	161
GUTHRIE	161
HALL STREET	138
HAMILTON	138
HUNTER	138
HUSTER	138
JOPPA 345	345
LAKESHIRE	138
LUTESVILLE	345
MALINE	138
MARION	138 & 161
MARSHALL	138
MAURER LAKE	161
MCCLAY	138
MINER	161
MOBERLY	161
PAGE	138
PHELPS	138
PIKE	161
PILOT KNOB	161
POINT PRAIRIE	161
RIDGE	138
RIVERMINES	138
SANDY CREEK	138

SELMA	138
SOUTH QUINCY	138

Substations - Switching and Bulk

Capotationo Cuitoning and Bank		
	For Specified Voltage Only	
SUBSTATION NAME	kV	
SPENCER CREEK	345	
STODDARD	161	
SWEETWATER	161	
VIBURNUM	161	
TEGELER	138	
WARRENTON	161	
WARSON	138	
WATSON	138	

\underline{kV}	Line Description
345	Ashtabula to Penelec Co. Tie
345	Avon (CEI) to Beaver #1
345	Avon (CEI) to Beaver #2
345	Avon to Beaver #2 (O.E. Tie)
345	Avon to Beaver (O.E. Tie)
345	Avon to Juniper
345	Bay Shore to Davis - Besse
345	Bay Shore to Monroe
345	Bay Shore to O.P.Tie-Fostoria
345	Beaver to Carlisle
345	Beaver to Davis-Besse(TE)
345	Beaver Val.(DL) to Hanna
345	Beaver Val.(DL) to Sammis
345	Canton Cen.(OP) to Hanna
345	Carlisle to Star
345	Chamberlin (CEI) to Harding
345	Chamberlin (CEI) to Mansfield
345	Davis - Besse to Lemoyne
345	Davis - Besse to O.E. Tie-Beaver
345	Eastlake Tap to Eastlake
345	Eastlake Tap to Juniper
345	Eastlake Tap to Nursery (T13 179)
345	Hanna to Highland
345	Hanna to Juniper (CEI)
345	Harding to Fox
345	Highland to Mansfield
345	Highland to Shenango
345	Hoytdale to Shenango
345	Hubbard, OH (1) to Shenango
345	Hyatt (OP) to Tangy
345	Inland to Harding
345	Juniper (CEI) to Star
345	Juniper to Canton (O.E. Tie)

345 Juniper to Chamberlin (O.E. Tie) 345 Juniper to Harding (Supply) 345 Juniper to Star (O.E. Tie) Lemoyne to Majestic 345 Lemoyne to Midway-Twr. 117 345 Lemoyne to OP. Tie-Fostoria 345 Mansfield (2) to Crescent 345 345 Mansfield (3) to Beaver Valley No. 1 345 Mansfield (4) to Highland 345 Mansfield (5) to Hanna-Harding Mansfield (6) to Beaver Valley No. 2 345 345 Mansfield Other Mansfield to Harding (DL Circuit #3 19) 345 Mansfield to Highland (DL Circuit #317) 345 345 Mansfield to Hoytdale 345 Mansfield to Hoytdale (DL Circuit # 313) 345 Marysville (OP) to Tangy 345 Midway-Twr. 117 to Allen Junction 345 Nursery (T13 179) to Perry 345 Perry to Ashtabula 345 Perry to Harding 345 Perry to Inland 345 South Canton (OP) to Star W.H.Sammis to Highland 345 W.H.Samniis to S. Canton (OP) 345 W.H.Samniis to Star 345 345 W.H.Sammis to Wylie Ridge Aeustar Tap No. ito Five Point 138 138 Acustar to 0.1. Levis Park Aetna: W. Akron to Aetna 138 138 Allen Jct. Tap No. 1 to Silica Allen Jct. to Jackman 138 138 Allen Jct. to Lyons

138 Allen Jct. to Reynolds 138 Allen Jct. to Sylvania American Steel to Bluebell 138 138 Angola to Hawthorne 138 Ashtabula to Ashtabula-C Ashtabula to PiUs.- Conn. Dock 138 138 Avery to Greenfield Avery to Shinrock 138 138 Avon to Dawson 138 Avon to Lorain 138 Ayersville to S.W. Defiance 138 BabbtoE.Akron 138 Babb to Valley 138 BabbtoW.Akron Barberton to Babcock-Wilcox 138 138 Barberton to Cloverdale Barberton to Star, North 138 138 Barberton to Star, South 138 Barberton to West Akron 138 Bay Shore Tap No. ito Hydra-Matic 138 Bay Shore Tap No. 2 to Hydra-Matic 138 Bay Shore Tap No. 3 to General Mills 138 Bay Shore to Decant 138 Bay Shore to Dixie No. 2 N.B. Bay Shore to Fort Industry 138 Bay Shore to Frey 138 Bay Shore to Ironville 138 138 Bay Shore to Jackman Bay Shore to Oregon 138 Beatty Rd.(C & SOE) to London 138 Beaver to Brookside 138 138 Beaver to Johnson 138 Beaver to NASA 138 Beaver to New Departure 138 Bellevue Tap No. ito UT No. 2 138 Bellevue to Vulcan No. 2 N.B.

Bluebell to Canton Cent.(OP)

138

138 Bluebell to Highland 138 Bluebell to Knox 138 Bluebell to Pidgeon Boardman to Riverbend 138 Boardman to Shenango 138 Brookside to Cloverdale 138 Brookside to Howard (OP) 138 138 Brookside to Leaside Brookside to Longview, East 138 138 Brookside to Longview, West Burger to Brookside 138 138 Burger to Cloverdale #1 138 Burger to Cloverdale #2 138 Burger to Cloverdale #3 138 Burger to Knox Burger to Longview 138 Cadillac to Ford 138 138 Canton Cent (OP) to Cloverdale 138 Carlisle to Gates 138 Carlisle to Johnson Carlisle to Lorain (CEI) East 138 138 Carlisle to Shinrock Central to Niles 138 138 Central to Packard Chamberlin to Darrow 138 138 Chamberlin to Valley 138 Chamberlin to West Akron 138 Clark to E. Springfield Clark to Greene 138 138 Clark to Urbana Cloverdale to East Wooster (OP) 138 138 Cloverdale to Star Cloverdale to Torrey (OP) 138 138 Crissinger to Roberts 138 Crissinger to Tangy 138 Crossland to Caparo 138 Crossland to Chemetron Corp.

Crystal to Avon-Fowles Tap 138 138 Dale to West Canton 138 Darrow to Hanna 138 Decant Tap No. ito Toussaint Decant Tap No. 2 to D. B. Mobile 138 Decant to Ottawa 138 138 Delaware (C&SOE Co.) to Tangy 138 DellTaptoDell 138 Delta to Wauseon 138 Dixie to Jackman No. 2 N.B. 138 E. Springfield to London 138 E. Springfield to Tangy East Akron to Gilchrist 138 138 EastAkrontoHarina East Akron to Sammis 138 East Akron to West Ravenna 138 138 Eastlake Tap to Eastlake 138 Eastlake to Nash 138 Eber to Liquid Air 138 Eber to Swanton 138 Edgewater to Beaver 138 Edgewater to USS (National Tube) Emily Tap to Emily 138 Emily to Galaxie 138 **Empire** 138 138 Evergreen to Franklin - Ivanhoe Evergreen to Highland #1 138 138 Evergreen to Highland #2 138 Evergreen to Highland #3 FayettetoStryker 138 Firestone to South Akron 138 Firestone to Urban 138 138 Ford 138 Ford Tap to Ford Motor Co. 138 Fort Industry Tap No. 1 to General Mills 138 Fort Industry to Jackman

Crossland to Sharon

138

Fowles to Clinton 138 138 Fowles to Dunbar Fowles to Galaxie 138 138 Fowles to NASA Fowles to Pleasant Valley 138 Frey to Walbridge Jct. 138 Galion to General Motors 138 138 Galion to Leaside Galion to Roberts, North 138 138 Galion to Roberts, South Garfield Tap to Garfield 138 138 Gates to Johnson 138 General Motors to Highland 138 General Motors to Longview 138 General Motors to Newton Falls Gilchrist to South Akron 138 GMC Tap to Gen. Motors Corp. 138 Greenfield to Lakeview 138 138 Greenfield to NASA 138 Greenfield to New Departure 138 Hanna to Shalersville Hanna to Newton Falls 138 Hanna to West Ravenna #1 138 138 Hanna to West Ravenna #2 Harding to Jennings (LTV East) 138 138 Hawthorne to Midway 138 Hazel to Horizon 138 Highland to Mahoningside 138 Highland to Salt Springs Hoytdale to Koppel 138 Hoytdale to Maple 138 138 Hoytdale to Pine **HunttoHunt** 138 138 Inca (T2456) to Linde 138 Inca (T2456) to Newburgh 138 Inland (T2600) to Ivy

Inland to Jordan

138

138 Ivanhoe to Mahoningside Ivanhoe to Packard 138 138 Ivy to Inland 138 Jackman to Bellevue No. 2 N.B. Jackman to Vulcan No. 1 N.B. 138 138 Jackman to Westgate Jennings to Clark 138 138 Jennings to Horizon 138 Jennings to J & L Steel Co.(LTW) 138 Jennings to Linde (Str. 241 Ox) 138 Johnson to Lorain (CEI) West 138 Johnson to USS (National Tube) 138 Juniper to Hillside-Inca (T2456) Juniper to Northfield 138 138 Juniper to Oak (12554) Juniper to Pleasant Valley 138 Keith Tap to Keith 138 138 Kendall to Kelly 138 Kenyon to Keith 138 Kepler to Kenyon 138 Kirby to Blue Jacket 138 Kirby to Roberts 138 Kirbytolangy Lakeshore to HL-HZ 138 Lakeview to Ottawa (TE) 138 138 Lark to Kepler 138 Lemoyne Tap No. 1 to Brim Lemoyne to Midway No. 1 138 138 Lemoyne to Midway No. 2 138 Lemoyne to O.P.Tie-Fost.No.1 138 Lemoyne to O.P.Tie-Fost.No.2 Lemoyne to Woodville No. 2 138 Leroy Center to Nursery 138 138 Leroy Center to Sanbom 138 Lester Tap to Lester 138 Liberty to Lark

138 Lincoln Park to Lowellville 138 Lincoln Park to Masury 138 Lloyd to Jordan 138 Lorain to Fowles Lorain to Johnson Tap 138 Lorain to Ohio ED. Co. Tie 138 Lowellville to Sanimis 138 138 Lyons to Fayette 138 MacLean Tap No. ito Walbridge Coatings 138 MacLean Tap No. 2 to Hunt Wesson 138 MacLean to Acustar 138 Maple to Connoquenessing Maple to Pine 138 Masury to Maysville (PP) 138 138 Masury to Salt Springs Masury to Sharon (PP)-Crosslands (PP) 138 138 Masury to Shenango (PP) 138 Masury, OH (1) to Crossland 138 Mayfield to Leroy Center 138 Mayfield to Lloyd 138 Midway Tap No.1 - Naomi Jct to Wauseon 138 Midway to Ridgeville Cor. No.2 138 Midway to Ridgeville Or. No. 1 Nash to Nursery 138 Nevada to Nevada 138 Niles to Bluebell 138 138 Niles to Evergreen 138 Niles to Salt Springs 138 Northfield to Mayfield 138 0.1. Levis Park Tap No.1 to Waterville 0.1. Levis Park Tap No.2 to Johns-Manville 138 0.1. Levis Park Tap No.3 to Johns-Manville 138 0.1. Levis Park to Midway 138 138 Orangeville, PA (1) to Maysville 138 Oregon to Walbridge Jct. 138 Ottawa Tap No. ito Kelsey-Hayes

Ottawa to 0.E. Tie-Lakeview

138

138 Ottawa to W. Fremont 138 Pidgeon to Sammis 138 Pinegrove Tap to Pinegrove Pleasant Valley (CEI) to W. Akron, East 138 Pleasant Valley (CEI) to W. Akron, West 138 Pleasant Valley to Jennings (P V-Hard) 138 138 Pleasant Valley to Ohio ED. Co. Tie Republic Stl. Tap to Republic Stl. Co. 138 Reynolds Tap No. 1 to UT No. 1 138 138 Reynolds to Vulcan Richland to Ayersville 138 138 Richland to G.M. East 138 Richland to G.M. West 138 Richland to O.P. Tie-E. Lima 138 Richland to O.P.Tie-Rob. Park 138 Ridgeville Cor. to Ridgeville No. 2 Ridgeville Cor. to Stryker 138 138 Ridgeville Cors. to Richland No. 1 138 Ridgeville to Richland 138 Riverbend to Salt Springs S.W. Defiance Tap No. 1 to Jobns-Manville 138 S.W. Defiance to Richland 138 Sammis to Boardman 138 138 Sanborn to Ashtabula 138 Shenango to Crossland 138 Shenango to Hubbard, OH (1) 138 Shenango to McDowell 138 Shenango to Sharon 138 Solon to Kendall 138 South Akron to Sunnyside-Torrey-W.Canton (OP) 138 Star to Urban 138 Star to West Akron 138 Star to West Medina 138 Swanton to Delta 138 Sylvania to Talmadge Talmadge to Westgate 138 Vulcan Tap No. ito Angola 138

138

Vulcan to Eber

- 138 Walbridge Jct. To Lemoyne No. 1
- 138 Walbridge Jct. To Lemoyne No. 2
- Walbridge Jct. To MacLean No. 1
- 138 Walbridge Jct. To MacLean No. 2
- Wellington
- 138 West Akron to West Medina #1
- 138 West Pittsburg to Hoytdale
- 138 West Pittsburg to Shenango
- Woodville No. 2 to W. Fremont
- 69 Trans Lines 69 kV Central OECO
- 69 Trans Lines 69 kV Eastern OECO
- 69 Trans Lines 69 kV PPCO
- 69 Trans Lines 69 kV Southern OECO
- 69 Trans Lines 69 kV Western OECO
- 69 Trans Lines 69 kV Western TECO

Substations

ABBE

ACME

ALLEN JUNCTION

ASHTABULA

AURORA

A VERY

AVON LAKE

BABB

BARBERTON

BAY SHORE

BEAVER

BEAVER VALLEY

BELLE VUE

BLUEBELL

BOARDMAN

BRIM

BROAD VIE W

BROOKSIDE

BURGER

CAMPBELL

CAPARO

CARLISLE

CARRIAGE

CEDAR STREET

CHAMBERLIN

CHITTENDEN CORNERS

CLARK

CLOVERDALE

COLUMBIANA

COOK

CROSSLAND

DALE

DARROW DAVIS-BESSE DETROIT DIXIE DUBLIN EAST AKRON EAST SPRINGFIELD EASTLAKE EDGE WATER ELL WOOD CITY ENON EVERGREEN FOWLES FOX FRISCO GALION GARRETTSVILLE GILCHRIST GOULD GREENFIELD GREENVILLE HANNA HARDING HAZEL HIGHLAND HOYTDALE INLAND IRON VILLE IVANHOE JACKMAN JENNINGS JOHNSON JUNIPER KNOX LAKE SHORE **LEASIDE LEMOYNE**

LEROY CENTER

LOWELLVILLE MACEDONIA MACLEAN MANSFIELD MAPLE MASURY MAYFIELD MAYSVILLE MCDOWELL MEDINA MIDWAY NEVADA NEW CASTLE NEWTON FALLS NILES OTTAWA PERRY PIDGEON PINE PLEASANT VALLEY **RAVENNA RICHLAND ROBERTS** SALT SPRINGS **SAMMIS SEVILLE SHALERSVILLE SHARON SHENANGO SHINROCK SOUTH AKRON STAR**

STRYKER

LIBERTY LOCUST LONDON LONG VIEW LORAIN

TANGY
TOUSSAINT
VULCAN
WAUSEON
WELLINGTON
WEST AKRON
WEST FREMONT
WEST RAVENNA
WOODFORD

FROM	<u>TO</u>	LINE #	<u>KV</u> <u>VOLTAGE</u>
DUMONT (AEP)	STILLWELL	34501	345
DUNE ACRES	BABCOCK	34502	345
BABCOCK	LAKE GEORGE	34503	345
MUNSTER	BURNHAM (CECO)	34504	345
MICHIGAN CITY GEN STA	BABCOCK	34505	345
MICHIGAN CITY GEN STA	DUNE ACRES	34506	345
SCHAHFER GEN STA	TOWER ROAD	34507	345
SCHAHFER GEN STA	LAKE GEORGE	34508	345
BURR OAK	LEESBURG	34509	345
DUNE ACRES	CHICAGO AVE	34510	345
SHEFFIELD	STATE LINE GEN STA (CECO)	34511	345
SHEFFIELD	BURNHAM (CECO)	34512	345
SCHAHFER GEN STA	BURR OAK	34513	345
BABCOCK	STILLWELL	34514	345
SHAHFER GEN STA	GREEN ACRES	34516	345
LEESBURG	DEEDSVILLE (CINERGY)	34518	345
SHEFFIELD	CHICAGO AVE	34520	345
SCHAHFER GEN STA	ST. JOHN	34521	345
LAKE GEORGE	MUNSTER	34522	345
TOWER ROAD	BABCOCK	34523	345
LEESBURG	HIPLE, F. G.	34524	345
MARKTOWN	WHITING CLEAN ENERGY PROJECT	13801*	138
BATAVIA (CONSUMERS ENERGY)	DEDICATED FACILITY BARTON LAKE	13802	138
STATE LINE GEN STA (CECO)	WOLF LAKE	13803	138
SHEFFIELD	CALUMET	13804	138
AETNA	DUNE ACRES & PRAXAIR, INC #5 – BURNS HARBOR	13805	138

FROM	<u>TO</u>	LINE #	<u>KV</u> <u>VOLTAGE</u>
AETNA	DUNE ACRES	13806	138
MONTICELLO	LAFAYETTE (CINERGY)	13807	138
ROXANA	LTV STEEL CO #2	13808	138
HIPLE, F. G.	LAGRANGE	13809	138
BURNS DITCH	MILLER	13810	138
CHICAGO AVE	PRAXAIR INC #1-EAST CHICAGO	13811	138
MAPLE	L.N.G. PLANT	13812	138
MICHIGAN CITY GEN STA	LAPORTE JCT (AEP) & OLIVE (AEP)	13813	138
MICHIGAN CITY GEN STA	TRAIL CREEK	13814	138
MICHIGAN CITY GEN STA	LUCHTMAN RD	13815	138
NEW CARLISLE (AEP)	MAPLE	13816	138
MILLER	US STEEL-TIN MILL	13817	138
AETNA	PRAXAIR INC #3-LAKESIDE	13818	138
BURR OAK	PLYMOUTH	13819	138
FLINT LAKE	STARKE	13820	138
PLYMOUTH	KOSCIUSKO & LEESBURG	13821	138
LAKE GEORGE	MILLER	13822	138
MUNSTER	HARTSDALE	13823	138
MARKTOWN	WHITING CLEAN ENERGY PROJECT	13824*	138
MITCHELL GEN STA	DEDICATED FACILITY ROXANA	13825	138
MITCHELL GEN STA	US STEEL-TIN MILL	13826	138
MICHIGAN CITY GEN STA	MAPLE	13827	138
DEKALB	AUBURN (AEP)	13828	138
CHICAGO AVE	INLAND #5	13829	138
MARKTOWN	INLAND #5	13830	138
CHICAGO AVE	PRAXAIR INC #3 – LAKESIDE	13831	138

FROM	<u>TO</u>	LINE #	<u>KV</u> <u>VOLTAGE</u>
EAST WINAMAC	MONTICELLO	13832	138
CHICAGO AVE	INLAND #7	13833	138
HARTSDALE	ST. JOHN	13834	138
ROXANA	PRAXAIR INC #1 – EAST CHICAGO	13835	138
DUNE ACRES	BURNS DITCH	13836	138
MITHICAN CITY GEN STA	DUNE ACRES	13839	138
MARKTOWN	LTV STEEL CO #2	13841	138
MILLER	BETA STEEL ARC FURNACE	13842	138
MICHIGAN CITY GEN STA	DUNE ACRES	13843	138
NORTHEAST	LEESBURG	13844	138
MITCHELL GEN STA	MITCHELL GEN STA	13845	138
REYNOLDS	MONTICELLO	13847	138
DUNE ACRES	BETHLEHEM STEEL CO	13848	138
DUNE ACRES	BETHLEHEM STEEL CO	13849	138
MARKTOWN	LTV STEEL CO #3	13850	138
SCHAHFER GEN STA	THAYER	13851	138
TRAIL CREEK	NEW CARLISLE (AEP)	13852	138
STATE LINE GEN STA	ROXANA	13853	138
(CECO) AETNA	LAKE GEORGE	13854	138
NORTHPORT	ALBION (AEP)	13856	138
GOODLAND	REYNOLDS	13857	138
MARKTOWN	INLAND #7	13858	138
CHICAGO AVE	US STEEL-STOCKTON	13859	138
MITCHELL GEN STA	US STEEL-COKE PLANT	13860	138
AETNA	US STEEL-WEST MILL	13861	138
LAKE GEORGE	TANEY	13862	138

FROM	<u>TO</u>	<u>LINE</u> #	<u>KV</u> <u>VOLTAGE</u>
LAKE GEORGE	HIGHLAND	13863	138
HENDRICKS	US STEEL-STOCKTON	13864	138
MILLER	US STEEL-COKE PLANT	13865	138
LAKE GEORGE	TOWER ROAD	13866	138
LAKE GEORGE	LIBERTY PARK	13867	138
ST. JOHN	LIBERTY PARK	13868	138
MARKTOWN	AMOCO	13869	138
ROXANA	CALUMET	13870	138
GOODLAND	WATSEKA (AMEREN)	13871	138
TOWER ROAD	FLINT LAKE	13872	138
FLINT LAKE	LUCHTMAN ROAD	13873	138
SCHAHFER GEN STA	STARKE & THAYER	13874	138
DUNE ACRESS	BABCOCK	13875	138
SHEFFELD	AMOCO	13877	138
SHEFFELD	MARKTOWN	13878	138
DUNE ACRES	BETA STEEL ARC FURNACE	13879	138
NORTHEAST	GOSHEN JUNCTION	13880	138
KOSCIUSKO	LEESBURG	13881	138
BURR OAK	EAST WINAMAC	13882	138
SOUTH PRAIRIE	WESTWOOD (CINERGY)	13883	138
DUNE ACRES	PRAXAIR INC #5-BURNS HARBOR	13884	138
LAKE GEORGE	AINSWORTH & GREEN ACRES	13885	138
SCHAHFER GEN STA	TOWER ROAD	13886	138
LAGRANGE	NORTHPORT	13887	138
GREEN ACRES	ST. JOHN	13888	138
HENDRICKS	US STEEL-WEST MILL	13889	138
CHICAGO AVE	INLAND #8	13890	138

FROM	<u>TO</u>	LINE #	<u>KV</u> <u>VOLTAGE</u>
MITCHELL GEN STA	INLAND #8	13891	138
MITCHELL GEN STA	CHICAGO AVE	13892	138
WOLF LAKE	SHEFFIELD	13893	138
MUNSTER	KENWOOD	13894	138
MUNSTER	TANEY	13895	138
PLYMOUTH	STILLWELL	13896	138
STILLWELL	L.N.G. PLANT	13897	138
TOWER ROAD	BABCOCK	13898	138
HIGHLAND	KENWOOD	13899	138
ST. JOHN	KREITZBURG	138-101	138
AETNA	MILLER	138-102	138
HIPLE, F. G.	GOSHEN JUNCTION	138-103	138
BALLY GEN STA-UNIT 7	DUNE ACRES	138-601	138
BALLY GEN STA-UNIT 8	DUNE ACRES	138-602	138
BALLY GEN STA-R.A.T.	DUNE ACRES	138-603	138
NORTHEAST	KLINE (AEP)	13855*	138
(* BREAKERS ONLY) NORTHEAST	* LINE OWNED BY AEP COLUMBIA (AEP)	13876*	138
(* BREAKERS ONLY) BURNS DITCH	* LINE OWNED BY AEP MIDWEST STEEL	13837*	138
(* BREAKERS ONLY) BURNS DITCH	* LINE OWNED BY MIDWEST STEEL MIDWEST STEEL	13838*	138
(* BREAKERS ONLY) ST. JOHN	* LINE OWNED BY MIDWEST STEEL EAST FRANKFORT (CECO) & CRETE	34519*	345
(* BREAKERS ONLY)	ENERGY PROJECT * LINE OWNED BY CECO	0.019	
ST. JOHN (* BREAKERS ONLY)	GREEN ACRES * LINE OWNED BY CECO	34517*	345
GREEN ACRES	OLIVE (AEP)	34515*	345
(* BREAKERS ONLY) LIBERTY PARK	* LINE OWNED BY CECO DEMOTTE	6901	69
	HEBRON HILLSIDE		
	MAPLEWOOD SW #416 TIE WITH 6930		
GOSHEN JCT	WAKARUSA	6902	69

FROM	<u>TO</u>	<u>LINE</u> #	<u>KV</u> <u>VOLTAGE</u>
FLINT LAKE	REXAM KANKAKEE VALLEY REMC-WASHINGTON SMITH DITCH PACTIV, INC WANATAH WANATAH WILLIAMSBURG SW #541 TIE WITH 6950	6903	69
PLYMOUTH	BASS LAKE CULVER KNOX KNOX JUNCTION ROCKWELL WINDOWS	6904	69
STARKE	NORTH JUDSON	6905	69
LIBERTY PARK	CEDAR LAKE COURT	6906	69
MONTICELLO	SW #386 TIE WITH DEAD LINE BUFFALO PIKE BURNETTSVILLE GUERNSEY IDAVILLE JORDAN MFG CO MONON SW #909 TIE WITH 6918	6907	69
LIBERTY PARK	LINCOLN SQUARE MERRILLVILLE SW #284 TIE WITH 6928 SW #823 TIE WITH 6953	6908	69
LIBERTY PARK	ASHLAND PRODUCTS CEDAR LAKE CRESTON HANOVER LAKE HILLS LOWELL RIETER AUTOMOTIVE-GLOBE INDUSTRIES SCHNEIDER TEPPCO – LOWELL PUMP STATION	6909	69
MONON	SALEM VULCAN MATERIALS WARD STONE CO	6910	69
SALEM	MOODY	6911	69
LIBERTY PARK	COURT MAPLEWOOD ORCHARD GROVE	6912	69
EAST WINAMAC	BRUCE LAKE WINAMAC	6913	69
PLYMOUTH	DONALDSON EAST WALKERTON HAMLET	6914	69

FROM	<u>TO</u>	LINE #	<u>KV</u> VOLTAGE
PLYMOUTH	MARSHALL CO REMC-TYNER SHILO CENTER HOOSIER TIRE & MFG. CORP. NORTH LIBERTY PIDCO SUNDOR BRANDS	6915	69
WINAMAC	SW #932 TIE WITH 6946 LINK	6916	69
KREITZBURG	DYER NOVAK ROAD PALMIRA	6917	69
MONTICELLO	SCHERERVILLE SW #781 TIE WITH 6932 COLLEGE LANDIS PLASTICS MILROY MONON	6918	69
STARKE	SW #366 TIE WITH 6991 SW #874 TIE WITH 6958 SW #909 TIE WITH 6907 SW #964 TIE WITH 6969 ASPHALTUM MEDARYVILLE PLYMOUTH TUBE CO SALEM	6919	69
LAKE GEORGE	WINAMAC HOBART RAND WILLOWDALE	6920	69
PLYMOUTH	SW #1099 TIE WITH 6954 SW #556 TIE WITH 6922 CENTER DEL MONTE CORP MARSHALL	6921	69
LAKE GEORGE	PACTIV, INC PLYMOUTH DEEP RIVER HOBART WHEELER	6922	69
HIPLE, F G	SW #1002 TIE WITH 6933 SW #556 TIE WITH 6920 LAKE INDUSTRIES OF INDIANA INC. MIDDLEBURY NEWBURY	6923	69
FLINT LAKE	LAGRANGE CO. REMC – BONTRAGER SW #281 TIE WITH 6976 EVANS MCGILL MFG CO #1 SUB SPECTACLE LAKE	6924	69

FROM	<u>TO</u>	LINE #	<u>KV</u> VOLTAGE
LUCHTMAN RD	LAKEHEAD PIPELINE CO-LAPORTE PINOLA SUMMIT	6925	69
GREEN ACRES	BROADWAY LINCOLN SQUARE MERRILLVILLE SOUTH LAKE	6926	69
TANEY	SW #421 TIE WITH 6930 BON AIRE INDEPENDENCE HILL ROSS SIXTY-FIRST AVE	6927	69
LAKE GEORGE	MISSISSIPPI SIXTY-FIRST AVE SW #284 TIE WITH 6908	6928	69
LAKE GEORGE	EAST GARY RAND	6929	69
GREEN ACRES	BROADWAY POLYCON INDUSTRIES SOUTH LAKE SW #416 TIE WITH 6901 SW #421 TIE WITH 6926	6930	69
SOUTH PRAIRIE	WEST POINT SW #131 TIE WITH 6967	6931	69
HARTSDALE	AMOCO PIPELINE-GRIFFITH LAKE HILLS LAKEHEAD PIPELINE CO-GRIFFITH SW #700 TIE WITH 6953 SW #781 TIE WITH 6917	6932	69
GREEN ACRES	DEEP RIVER SOUTH HAVEN TRANSMONTAIGNE PIPELINE -WHEELER WHEELER SW #1002 TIE WITH 6922 SW #597 TIE WITH 6955	6933	69
GREEN ACRES	BUCKEYE PIPELINE CO-KOUTS CORNELL HEBRON KANKAKEE VALLEY REMC-PORTER SUB KANKAKEE VALLEY REMC-JACOB MACK MALDEN MERIT STEEL CO	6934	69
THAYER	SW #941 TIE WITH 6939 LOWELL ORCHARD GROVE SCHNEIDER SW #1310 TIE WITH 6958 SW #691 TIE WITH DEAD WIRE	6935	69
NORTHPORT	LAGRANGE CO REMC-INDIANA LAKES WOLCOTTVILLE	6936	69

FROM	<u>TO</u>	<u>LINE</u> #	<u>KV</u> VOLTAGE
EAST WINAMAC	BRUCE LAKE CULVER SW #853 TIE WITH 6938	6937	69
CULVER	WALKER MFG CO SW #853 TIE WITH 6937	6938	69
GREEN ACRES	KANKAKEE VALLEY REMC-JACOB MACK SW #941 TIE WITH 6934	6939	69
MAPLE	AMERICAN RENOLIT FISH LAKE FOAMEX CORP SILGAN CONTAINER CORP - LAPORTE SUPERIOR REAL ESTATE SW #391 TIE WITH 6960 SW #406 TIE WITH 6978 SW #890 TIE WITH 6945	6940	69
NORTHEAST	MILFORD NEW PARIS SYRACUSE WARNER RD. SW #1315 TIE WITH 6999	6941	69
ANGOLA	FREMONT HOOSIER HILL INDIANA WIRE CO. JAMES RITTAL ELECTROMATE STEUBEN CO REMC-SCOTT WHEEL-TEK BARTON LAKE SW #923 TIE WITH 6994	6942	69
LAGRANGE	DOMETIC CORP. HOWE LAGRANGE CO REMC-PIGEON LAKE NEWBURY SPRING	6943	69
THIRD ST	BREMEN CITY I	6944	69
STILLWELL	CASTEEL SERVICES INC DEER RUN KINGSBURY ACCURATE CASTINGS CORP – KINGSBURY ROLL COATER, INC	6945	69
STILLWELL	SW #890 TIE WITH 6940 EAST WALKERTON KANKAKEE VALLEY REMC-J. BOWMAR NORTH LIBERTY SHILO WELLS BROTHERS ALUMINUM SW #932 TIE WITH 6915	6946	69
NORTHEAST	DOME PIPELINE CORP KOSCIUSKO CO REMC-MAPLE LEAF	6947	69

FROM	<u>TO</u>	<u>LINE</u> #	<u>KV</u> <u>VOLTAGE</u>
BABCOCK	MILFORD MONOQUET OSWEGO SW #1165 TIE WITH 6975 CHESTERTON CHESTERTON HIGH SCHOOL MAGNETIC INTERNATIONAL INC MCCOOL PROCESSING TECHNOLOGY INC TOWNSHIP	6948	69
FLINT LAKE	SW #1116 TIE WITH 6955 BURDICK ROAD KANKAKEE VALLEY REMC-WEST TOLL PURDUE UNIVERSITY-NORTH CENTRAL WESTVILLE	6949	69
FLINT LAKE	INDIANA DEPT. OF CORRECTION- WESTVILLE ISK MAGNETICS ROUNDYS, INC. POWDERTECH CORP WESTVILLE	6950	69
KREITZBURG	SW #541 TIE WITH 6903 CRESTON HANOVER	6951	69
FLINT LAKE	EVANS GRAND TRUNK UGIMAG INC URSCHEL LABORATORIES INC WASHINGTON	6952	69
LIBERTY PARK	SW #1060 TIE WITH 6956 INDEPENDENCE HILL LAKE CO GOVERNMENT COMPLEX SW #700 TIE WITH 6932 SW #823 TIE WITH 6908	6953	69
BABCOCK	MCCOOL PORTAGE HIGH SCHOOL SOUTH HAVEN WILLOWDALE WOODLAND PARK SW #1099 TIE WITH 6920	6954	69
BABCOCK	GRAND TRUNK SPECTACLE LAKE SW #1116 TIE WITH 6948 SW #597 TIE WITH 6933	6955	69
FLINT LAKE	AOC RESINS AND COATING MALDEN MCGILL MFG CO #2 SUB JET CORP. SMITH DITCH UNION ELECTRIC STEEL	6956	69

FROM	<u>TO</u>	LINE #	<u>KV</u> VOLTAGE
	VALPARAISO UNIVERSITY WASHINGTON WILLIAMSBURG SW #1060 TIE WITH 6952		
HARTSDALE	PLUM CREEK AVERY DENNISON SCHERERVILLE	6957	69
THAYER	AMOCO DEMOTTE PUMP STA DEMOTTE EGER JASPER CO REMC-FAIR OAKS JASPER CO REMC-KERSEY JASPER CO REMC-SUTTON PARR SW #874 TIE WITH 6918 SW #1310 TIE WITH 6935	6958	69
ANGOLA	HELMER HUDSON NOBLE COUNTY REMC – FAIRFIELD SOUTH MILFORD WOLCOTTVILLE SW #376 TIE WITH 6986	6959	69
MAPLE	DIVISION GREENWAY HOWMET CORP TELEDYNE CASTING SERVICE WHIRLPOOL CORP SW ##30 TIE WITH 6983 SW #391 TIE WITH 6940	6960	69
STARKE	HAMLET HANNA KANKAKEE VALLEY REMC-ARROW KANKAKEE VALLEY REMC-DOME PIPELINE CO KINGSFORD HEIGHTS KNOX WANATAH	6961	69
EAST WINAMAC	BASS LAKE LAWTON NORTH JUDSON	6962	69
GOODLAND	GOODLAND JCT KENTLAND ROGERS GROUP INC.	6963	69
GOODLAND	FOWLER GOODLAND JCT KENTLAND PINE CREEK WARREN CO REMC-DUNNINGTON SW #134 TIE WITH 6967	6964	69

FROM	<u>TO</u>	LINE #	<u>KV</u> VOLTAGE
KENTLAND	CAPITAL PRODUCTS MOROCCO VISKASE CORP	6965	69
GOODLAND	CENTRAL SOYA JASPER CO REMC-CARPENTER OMNI FORGE REMINGTON	6966	69
GOODLAND	FOWLER PINE CREEK WARREN CO REMC-TEMPLETON	6967	69
PLYMOUTH	SW #131 TIE WITH 6931 SW #134 TIE WITH 6964 ARGOS ARGOS CITY-DEWEY ST ARGOS INDUSTRIAL PARK MENTONE ROCHESTER (CINERGY)	6968	69
GOODLAND	SW #1031 TO DEAD WIRE BANET BROOK MOROCCO	6969	69
GOSHEN JCT	SW #964 TIE WITH 6918 DAIRY FARM PRODUCTS ILLINOIS ROCK RUN WESTERN CONSOLIDATED TECH	6970	69
MONTICELLO	OAKDALE HYDRO TWIN LAKES	6971	69
OAKDALE HYDRO	SOUTH CHALMERS SOUTH PRAIRIE WHITE CO REMC-ULERICH	6972	69
LAGRANGE	SPRING	6973	69
PLYMOUTH	BOURBON BREMEN CITY I IRONWOOD LEAR CORP THIRD ST	6974	69
NORTHEAST	BOURBON CLUNETTE NAPPANEE VITCO INC SW #1162 TIE WITH 6981 SW #1165 TIE WITH 6947	6975	69
GOSHEN JCT	SW #6 TIE WITH 6996 ABTCO BONNEYVILLE CRYSTAL VALLEY MIDDLEBURY ROCK RUN	6976	69

FROM	<u>TO</u>	LINE #	<u>KV</u> VOLTAGE
GOSHEN JCT	SYNDICATE STORE FIXTURES INC SW #281 TIE WITH 6923 BONNEYVILLE BRISTOL HAYES-LEMENZ HINSDALE FARMS, LTD.	6977	69
MAPLE	MIDWAY MODEL LAPORTE JCT (AEP) SPRINGVILLE (AEP) SUMMIT WOLVERINE PIPELINE CO - LAPORTE SW #406 TIE WITH 6940	6978	69
MONTICELLO	NORWAY HYDRO	6979	69
LAGRANGE	ANGOLA	6980	69
GOSHEN JCT	OTTER SW #160 TIE WITH 69-102 SW #666 TIE WITH 69-100 BREMEN CITY II NAPPANEE NORTHWOOD THIRD ST WAKARUSA	6981	69
KOSCIUSKO	SW #1162 TIE WITH 6975 KOSCIUSKO COUNTY REMC-AIRPORT MONOQUET NORTH WEBSTER OSWEGO PIERCETON SPARTECH	6982	69
MAPLE	SW #582 TIE WITH 6995 ACCURATE CASTINGS INC - LAPORTE FIELD CONTAINER CORP KEYSTONE SERVICES LAPORTE SW #30 TIE WITH 6960 SW #532 TIE WITH 6988	6983	69
GOSHEN JCT	MODEL	6984	69
GOSHEN JCT	BENTELER INDUSTRIES GOSHEN RUBBER CO HORN DITCH ILLINOIS NIBCO PINE MANOR SW #1250 TIE WITH 6990	6985	69
DEKALB	ANGOLA ASHLEY HELMER	6986	69

FROM	<u>TO</u>	<u>LINE</u> #	<u>KV</u> VOLTAGE
	STEUBEN CO REMC-EAST ANGOLA WATERLOO STEUBEN CO REMC - WATERLOO INDUSTRIAL PARK		
LAGRANGE	SW #376 TIE WITH 6959 HOWE LAGRANGE CO. REMC-NORTH LAGRANGE	6987	69
MAPLE	GREENWAY JEFFERSON SMURFIT CORP. KINGSBURY LAPORTE PINOLA	6988	69
NORTHEAST	SW #532 TIE WITH 6983 DANA CORP-SYRACUSE NORTH WEBSTER SYRACUSE SYRACUSE RUBBER CO #2 WAWASEE	6989	69
HIPLE, F G	BUTTERMILK CORNERS NORTHPORT TOPEKA SW #1250 TIE WITH 6985	6990	69
MONTICELLO	BALL CORP. HONEY CREEK MCGILL MFG CO-MONTICELLO SHARP STEEL SERVICES SW #366 TIE WITH 6918	6991	69
KOSCIUSKO	DALTON FOUNDRIES INC MCKINLEY WARSAW SW #999 TIE WITH 6995	6993	69
ANGOLA	BARTON LAKE HOOSIER HILL JAMES UNIVERTICAL CORP SW #923 TIE WITH 6942 SW #1300 TIE WITH 69-102	6994	69
KOSCIUSKO	DEPUY INC. KOSCIUSKO CO REMC-EAST THIRTY SUB MCKINLEY PIERCETON WOOSTER SW #582 TIE WITH 6982 SW #999 TIE WITH 6993	6995	69
KOSCIUSKO	SW #6 TIE WITH 6975	6996	69
KOSCIUSKO	CLAY KOSCIUSKO CO REMC - KRALIS MENTONE R R DONNELLY & SONS	6997	69

FROM	<u>TO</u>	LINE #	<u>KV</u> <u>VOLTAGE</u>
	ROZELLA		
	WARSAW		
	WAYNE		
	WEIRCK		
	SW #1180 TIE WITH 6998		
	SW #1221 TIE WITH 6998	40.00	
KOSCIUSKO	FLINT INK CORP.	6998	69
	R R DONNELLY & SONS		
	WARSAW		
	WAYNE		
	ZIMMER MFG CO SW #1180 TIE WITH 6997		
	SW #1180 TIE WITH 6997 SW #1221 TIE WITH 6997		
GOSHEN JCT	NEW PARIS	6999	69
GOSTIEN JCT	PINE MANOR	0999	09
	SW #1315 TIE WITH 6941		
	SW #450 TIE WITH DEAD WIRE		
LAGRANGE	ANGOLA	69-100	69
2.10141,02	LAGRANGE CO REMC-PRAIRIE HEIGHTS	0, 100	
	SW #666 TIE WITH 6980		
HOWE	STURGIS (AEP)	69-101	69
LAGRANGE	BARTON LAKE	69-102	69
	BRIGHTON		
	FEDERAL MOGUL		
	NEVADA MILLS		
	STEUBEN CO REMC – NEVADA		
	SW #160 TIW WITH 6980		
SOUTH PRAIRIE	HONEY CREEK	69-103	69
	REMINGTON		

APPENDIX B

THIRD-PARTY JOINT AGREEMENTS

[Each Party shall set forth in a subappendix to this Appendix D all third-party joint agreements, if any, which are exceptions to such Party's representation and warranty set forth in Section 10.13.]

APPENDIX C NOTICES

GridAmerica LLC

GridAmerica LLC [Address]

Attn: Managing Member

Fax: **Ameren**

Ameren Services Company One Ameren Plaza 1901 Chouteau Avenue

St. Louis, MO 63103

Attn: David A. Whiteley

Senior Vice President

Fax: 314-554-4084

Ameren Services Company

One Ameren Plaza 1901 Chouteau Avenue St. Louis, MO 63103

Attn: Steven R. Sullivan

General Counsel

Fax: 314-554-4084

ATSI

American Transmission Systems, Incorporated c/o FirstEnergy Service Company South Main Street Akron, OH 44308

Attn: Stanley F. Szwed Fax: 330-384-4988

NIPSCO

Northern Indiana Public Service Company

801 E. 86th Avenue Merrillville, IN 46410

Attn: Jerry L. Godwin Fax: 219-647-5533

APPENDIX D SCHEDULE 5A - DELINEATION OF FUNCTIONS BETWEEN MIDWEST ISO, GRIDAMERICA AND GRIDAMERICA THREE

Responsibility	RTO	GridAmerica	Functions performed by MIDWEST ISO	Functions performed by GridAmerica under contract to MIDWEST ISO	Functions performed by GridAmerica	Functions performed by individually by GridAmerica Three in their role as Transmission Owner and Control Area Operator
Short term operational planning (including Maintenance Outages)	RTO approves maintenance for critical transmission facilities.	Coordinates maintenance of generators and non-critical transmission facilities in its area.	Identifies all critical transmission facilities for MIDWEST ISO footprint. Approves all outages except generator and non-critical transmission facilities in GridAmerica.	Optimizes and prescreens maintenance outages of critical transmission facilities in GridAmerica to ensure coordination with outages of transmission facilities outside the footprint so that regional reliability is met in the most economic manner.	Submits maintenance outages of all critical transmission facilities in GridAmerica for MIDWEST ISO approval. Approves maintenance outages of all non-critical transmission facilities and coordinates maintenance outages of all generator facilities in GridAmerica.	Provide Grid America with outage schedule proposals. Work with GridAmerica to understand/alleviate outage scheduling issues. Implement approved transmission facility outages and coordinate generation facility outages as directed by GridAmerica.
Operational Authority	Operates into, out of, and through transactions.	Schedules and physically operates transmission with source and sink inside footprint.	Ultimate approval authority as the Transmission Provider for all schedules in MIDWEST ISO footprint. Will have ultimate authority for any disputes with approvals or schedules.	Prescreens all schedules with source or sink in GridAmerica as Scheduling Entity. Coordinates actions with MIDWEST ISO and will respect any impacts on MIDWEST ISO flowgates.	Approves and implements all schedules with source and sink in GridAmerica as a Scheduling Entity for the Control Areas in the GridAmerica footprint.	Approves e-Tags that source and/or sink in CA. Implement schedules that source and/or sink in CA.
Reliability, Security and Coordination	Responsible for reliability for entire region.	Takes corrective action for reliability inside footprint under RTO supervision.	Monitors all critical transmission facilities in MIDWEST ISO footprint and oversees or directs corrective action in accordance with NERC policy. Monitors and approves GridAmerica actions.	Determine methods by which critical transmission facilities in GridAmerica footprint can be operated in the most economic manner to avoid transmission curtailment under fault conditions and recommends corrective action to ensure compliance with NERC policy.	Implements corrective action in GridAmerica under MIDWEST ISO supervision.	Monitors Control Area (CA) and provide advice to GridAmerica on corrective actions in CA which will be provided to RA as necessary. Implements corrective actions as instructed by GridAmerica and under the direction of GridAmerica and perform regional re-dispatch as required under current agreements (eg. LEER). Perform security analysis for CA/local TO facilities with "local model" and identify potential constraints to GridAmerica and provide advice in determining operating procedures or actions to alleviate such constraints.
Parallel Path Flows	Manages parallel path flow for region.	Assists in the management of parallel path flows during emergencies.	Monitors all critical transmission facilities for parallel path flow and provides remedial action when needed. Directs GridAmerica in management during emergencies.	Monitors all critical transmission facilities in GridAmerica footprint for parallel path flow and recommends corrective action.	Monitors all non-critical transmission facilities in GridAmerica footprint for parallel path flow. Implements corrective action in GridAmerica under MIDWEST ISO supervision.	Implements actions for parallel path flow remediation when directed by GridAmerica.
Congestion Management	Responsible for implementing	No responsibilities authorized at this time.	Develops and implements market based congestion management	Assists Midwest ISO in development of market based		

Responsibility	RTO	GridAmerica	Functions performed by MIDWEST ISO	Functions performed by GridAmerica under contract to MIDWEST ISO	Functions performed by GridAmerica	Functions performed by individually by GridAmerica Three in their role as Transmission Owner and Control Area Operator
	congestion management.		structure for all MIDWEST ISO footprint.	congestion management and incentive rate structures.		
Ancillary Services	Provider of last resort for ancillary services other than scheduling, system control and dispatch; voltage control; and regulation.	Provide ancillary services, scheduling, system control and dispatch; voltage control; and regulation service. GridAmerica may provide non-real time imbalance energy and ancillary services upon a showing of no harm to an RTO's ancillary service and imbalance energy markets.	Validates provision of ancillary services by GridAmerica to meet tariff requirements.	Periodically audits provision of ancillary services by third parties in GridAmerica footprint and recommends corrective action.	Proposes and implements approved provision of ancillary services allowed. Any financial obligations that are in addition to standard tariff terms will be borne by GridAmerica in accordance with any contract terms that are negotiated	Provide third party or POLR ancillary services on behalf of GridAmerica. Has ability to self-provide for bundled load. Will provide imbalance calculation (ancillary 4) for billing purposes until a regional market is adopted.
Planning and Expansion	Authority for region. Directs expansions as required. Develop joint planning protocol.	Develop joint planning protocol. Responsible for planning and expansion of its own system, but where RTO has ultimate authority when there are material impacts outside of GridAmerica.	Approves GridAmerica planning in GridAmerica footprint when plans have material affect in non-GridAmerica MIDWEST ISO footprint.	Assist Midwest ISO in developing a regional plan that improves the capability of the transmission system (and thereby facilitate efficient markets) by identifying engineering solutions (both to existing and new) in ensuring that secure and economically efficient plans are developed. Administer and actively process interconnection requests for the GridAmerica footprint. Analyzes and makes recommendations to coordinate planning in respect of ITCs under common management with GridAmerica in other RTOs.	Develop plans in GridAmerica footprint.	Provide GridAmerica with input/solutions to GridAmerica expansion plans and reliability plans, based on local knowledge.
Market Monitoring	Monitors market for entire region.	No market monitoring duties requested. Proposal to impose and collect penalties is rejected without prejudice.	Provides independent market monitoring in MIDWEST ISO footprint.			
Losses	Single method/system.	No responsibilities authorized at this time.	MIDWEST ISO provides losses for application in scheduling and settlement by GridAmerica in GridAmerica footprint.		Uses MIDWEST ISO provided losses for scheduling and settlement in GridAmerica footprint.	Provide GridAmerica with underlying data used as the basis for determining loss factors used for calculating losses.